On the Structure of Set-Theoretic Polygon Equations

Folkert MÜLLER-HOISSEN

Institut für Theoretische Physik, Friedrich-Hund-Platz 1, 37077 Göttingen, Germany E-mail: folkert.mueller-hoissen@phys.uni-goettingen.de URL: http://mueller-hoissen.math-phys.org

Received December 29, 2023, in final form May 29, 2024; Published online June 11, 2024 https://doi.org/10.3842/SIGMA.2024.051

Abstract. Polygon equations generalize the prominent pentagon equation in very much the same way as simplex equations generalize the famous Yang-Baxter equation. In particular, they appeared as "cocycle equations" in Street's category theory associated with oriented simplices. Whereas the (N - 1)-simplex equation can be regarded as a realization of the higher Bruhat order B(N, N - 2), the N-gon equation is a realization of the higher Tamari order T(N, N-2). The latter and its dual $\tilde{T}(N, N-2)$, associated with which is the dual N-gon equation, have been shown to arise as suborders of B(N, N - 2) via a "threecolor decomposition". There are two different reductions of T(N, N-2) and $\tilde{T}(N, N-2)$, to T(N-1, N-3), respectively $\tilde{T}(N-1, N-3)$. In this work, we explore the corresponding reductions of (dual) polygon equations, which lead to relations between solutions of neighboring (dual) polygon equations. We also elaborate (dual) polygon equations in this respect explicitly up to the octagon equation.

Key words: polygon equations; simplex equations, cocycle equations; pentagon equation; set-theoretic solutions; higher Bruhat orders; higher Tamari orders

2020 Mathematics Subject Classification: 06A06; 06A07; 15A69; 16T05; 16T25; 17A01; 18D10

1 Introduction

An infinite family of "polygon equations" has been introduced in [13], based on the combinatorial structure of higher Tamari (partial) orders. They appeared earlier as cocycle equations ("higher cocycloids") in a category-theoretical framework based on oriented simplices ("orientals") [57]. The 4-gon equation is the condition of associativity for a binary operation. The dual 4-gon equation expresses coassociativity of a comultiplication. The 5-gon or pentagon equation is the most prominent member of this family and plays a profound role in mathematics and mathematical physics.

Of particular relevance is the pentagon equation in the context of bi- and Hopf algebras. The crucial and fairly simple observation here is the following. Given a unital associative algebra \mathcal{A} with identity element $1_{\mathcal{A}}$, it carries a trivial (left) comultiplication¹ $\Delta_{\ell} : \mathcal{A} \to \mathcal{A} \otimes \mathcal{A}, a \mapsto a \otimes 1_{\mathcal{A}}$. Non-trivial comultiplications are then obtained via $\Delta := W \Delta_{\ell} W^{-1}$ if $W \in \mathcal{A} \otimes \mathcal{A}$ is an invertible solution of the pentagon equation² $W_{12}W_{13}W_{23} = W_{23}W_{12}$ in the threefold tensor product of \mathcal{A} , where the indices determine at which two components of it W acts. In fact, each comultiplication on a finite-dimensional algebra can be expressed in this way [48]. But also in the rigorous framework of infinite-dimensional C^* -algebras such an expression for comultiplication plays a profound role in surpassing technical problems in the duality theory of locally compact groups and for establishing a rigorous setting for quantum groups. Here W appears as a unitary

¹Of course, there is a corresponding result for the "right" trivial comultiplication $\Delta_r : \mathcal{A} \to \mathcal{A} \otimes \mathcal{A}, a \mapsto 1_{\mathcal{A}} \otimes a$.

²A somewhat more general result can be found in [9] and involves a "mixed" or "entwining" pentagon equation. Corresponding versions of polygon equations, involving several different maps, have been treated in full generality in [13]. Also see [32], for example.

linear operator under the name "multiplicative unitary" or (generalized) Kac–Takesaki operator (see, in particular, [1, 38, 54, 60, 61, 62, 69]).

The pentagon equation, sometimes called "fusion equation", shows up in conformal field theory as a condition for the fusion matrix, which arises from operator product expansion of chiral vertex operators, see [49] for example.

The pentagon equation plays a crucial role as an associativity constraint for the basic functor of a monoidal category, notably because of Mac Lane's coherence theorem [40] (also see [33]). This in turn has roots in Tamari's work on relaxing associativity and Stasheff's work on homotopy associativity [55, 56].

One way to obtain topological invariants of manifolds is via triangulations, i.e., decompositions into simplices. To each simplex one assigns a mathematical object and builds from all of them an object (e.g., a "state sum") assigned to a triangulation. This has to be done in such a way that the latter object is invariant under manipulations of triangulations called Pachner moves or bistellar flips [50] (also see, e.g., [39]). In three dimensions, the simplices are tetrahedra. A Pachner (2,3)-move splits two adjacent tetrahedra into three and this requires that the objects associated with the tetrahedra, constituting a triangulation, satisfy a pentagon relation. Because of the pentagonal Biedenharn–Elliot identity [4], a Wigner 6-*j* symbol, or a generalization of it, is thus a choice for the object assigned to a tetrahedron [51, 63, 64].³ In [3], it was shown that the Wheeler–DeWitt equation for three-dimensional general relativity reduces to the pentagon relation. There is a comprehensive literature around the idea of representing Pachner moves of a triangulation in three dimensions by a solution of the pentagon equation (or a pentagon relation), see in particular [2, 8, 24, 34, 47, 59]. In four dimensions, the dual hexagon equation plays a corresponding role [29] (also see Remark 6.2 below), in five dimensions it is the dual heptagon equation [35, 36, 37].

In the latter context, it is also worth to mention that the higher Tamari orders, introduced in [13] and underlying polygon equations, are equivalent [67] to higher Stasheff–Tamari orders [16, 23] on the set of triangulations of a cyclic polytope.

A pentagonal relation is also satisfied by the Rogers [52] and quantum dilogarithms⁴ [18, 19]. Although it does not have the structure of the pentagon equation as considered in this work, there are certain relations, see, e.g., Example 7.20 below.⁵

Comparatively little is known so far about higher polygon equations. An expression for the dual hexagon equation appeared as a 4-cocycle condition in [57, 58].

Polygon equations play a role in the theory of completely solvable models of statistical mechanics. If solutions of the N-gon and the dual N-gon equation satisfy a certain compatibility condition, a special solution of the (N - 1)-simplex equation is obtained, see [13, Section 5]. This includes and generalizes a relation between solutions of the pentagon equation, its dual, and the 4-simplex (Bazhanov–Stroganov) equation [31]. There is also a relation between the pentagon equation, its dual, and the 3-simplex (tetrahedron or Zamolodchikov) equation [31, 32, 41, 53], for which a corresponding generalization to higher polygon and simplex equations is yet unknown.

A class of solutions of odd polygon equations with maps acting on direct sums has recently been obtained in [10].

In the present work, we explore polygon equations in the set-theoretic setting, i.e., we do not assume any additional structure beyond sets and maps between (Cartesian products of)

³Essentially, (generalized) 6-*j* symbols arise as follows. If there is a direct sum decomposition $V_i \otimes V_j = \bigoplus_{\ell} H_{ij}^{\ell} \otimes V_{\ell}$ of vector spaces, the associator isomorphism $(V_i \otimes V_j) \otimes V_k \to V_i \otimes (V_j \otimes V_k)$ of a monoidal category induces maps $\begin{cases} i & j & \ell \\ k & m & n \end{cases}$: $H_{ij}^{\ell} \otimes H_{\ell k}^m \to H_{in}^m \otimes H_{jk}^n$.

⁴Also the quantum dilogarithm can be understood as a 6-j symbol [24].

⁵Also Drinfeld associators of quasi-Hopf algebras (see, e.g., [17]) are subject to a certain pentagon equation, but they are elements of a triple tensor product and therefore the corresponding pentagon equation is different from the standard one.

them. Using the combinatorics of higher Tamari orders, underlying polygon equations, we derive relations between the sets of solutions of neighboring (dual) polygon equations. These relations are proved for the whole infinite family of (dual) polygon equations. Additional results are provided for (dual) polygon equations up to the octagon (8-gon) equation.

Section 2 briefly recalls the definition of higher Bruhat and higher Tamari orders. The reader is referred to [13] for a detailed treatment. Section 3 presents the definition of simplex and polygon equations in a slightly different way from [13]. Section 4 deals with reductions of polygon equations and contains the most general results of this work. Sections 5 and 6 present the first few examples of polygon equations and their duals, respectively. They extend results obtained in [13].

Section 7 deals with the special, but most important case of (dual) polygon equations for a single map, acting between Cartesian products of a set \mathcal{U} . Partly from the general results in Section 4, but to some extent also independently and from results in Sections 5 and 6, we derive relations between solutions of a (dual) N-gon and (dual) (N + 1)-gon equation, N > 2.

Section 8 contains some concluding remarks.

2 Basics of higher Bruhat and higher Tamari orders

For a non-empty finite subset M of \mathbb{N} , and $n \in \mathbb{N}$, $1 \leq n \leq |M|$ (where |M| is the cardinality of M), let $\binom{M}{n}$ denote the set of *n*-element subsets of M. The *packet* P(M) of M is the set of (|M| - 1)-element subsets of M. We write $\overline{P}(M)$ for P(M) in lexicographic order, and $\overline{P}(M)$ for P(M) in reverse lexicographic order.

Let $N \in \mathbb{N}$, N > 1, and $[N] = \{1, \ldots, N\}$. A linear order (permutation) ρ of $\binom{[N]}{n}$, $n \in \mathbb{N}$, n < N, is called *admissible* if, for each $K \in \binom{[N]}{n+1}$, the packet P(K) is contained in ρ in lexicographic or in reverse lexicographic order. Let A(N, n) denote the set of admissible linear orders of $\binom{[N]}{n}$. An equivalence relation is defined on A(N, n) by $\rho \sim \rho'$ if and only if ρ and ρ' only differ by exchange of two neighboring elements, not both contained in some packet. Then $A(N, n)/\sim$, supplied with the partial order given via inversions of lexicographically ordered packets, $\overrightarrow{P}(K) \mapsto \overrightarrow{P}(K)$, is the higher Bruhat order B(N, n).

Next we consider the splitting of a packet, $P(K) = P_{o}(K) \cup P_{e}(K)$, where $P_{o}(K)$ ($P_{e}(K)$) is the half-packet consisting of elements with odd (even) position in the lexicographically ordered P(K).

We say an element $J \in P_{o}(K)$ is blue in $\overrightarrow{P}(K)$ and red in $\overleftarrow{P}(K)$, an element $J \in P_{e}(K)$ is red in $\overrightarrow{P}(K)$ and blue in $\overrightarrow{P}(K)$. $J \in P(K)$ is blue (red) in $\rho \in A(N, n)$ if J is blue (red) with respect to all K for which $J \in P(K)$ and either $\overrightarrow{P}(K)$ or $\overleftarrow{P}(K)$ is a subsequence of ρ .

It can happen that J is blue in $\rho \in A(N, n)$ with respect to some K and red with respect to another K'. In such a case we color it green. By $\rho^{(b)}$, $\rho^{(r)}$, $\rho^{(g)}$ we denote the blue, red, respectively green subsequence of ρ .

It has been shown in [13] that there are projections $B(N,n) \to B^{(c)}(N,n)$, $[\rho] \mapsto [\rho^{(c)}]$, $c \in \{b, r, g\}$, such that $B^{(c)}(N,n)$ inherits a partial order from B(N,n). $T(N,n) := B^{(b)}(N,n)$ are the higher Tamari orders and $\tilde{T}(N,n) := B^{(r)}(N,n)$ are called dual higher Tamari orders. The inversion operation in case of T(N,n) is $P_{o}(K) \mapsto P_{e}(K)$, $K \in {[N] \choose n+1}$. In case of $\tilde{T}(N,n)$, it is $P_{e}(K) \mapsto P_{o}(K)$.

Remark 2.1. Associating with $K \in {[N] \choose n+1}$ an *n*-simplex, the packet P(K) corresponds to the set of its faces, which are (n-1)-simplices. The pasting scheme given by the above inversion operation then supplies the faces with an orientation. This results in the orientals (oriented simplices) introduced by Street in 1987 [57]. It had been conjectured in [12, 13] and proved in [67] that the higher Tamari orders are equivalent to the higher Stasheff-Tamari orders in [16, 23]. All these works thus deal with essentially the same structure. In [11, 12] it has been realized in terms of rooted tree-shaped solutions of the Kadomtsev-Petviashvili (KP) hierarchy.

3 A brief account of simplex and polygon equations

Let N > 2. With $J \in {[N] \choose N-2}$ we associate a set \mathcal{U}_J . For $\rho \in A(N, N-2)$, let \mathcal{U}_ρ be the correspondingly ordered Cartesian product of the \mathcal{U}_J , $J \in \rho$. With $K \in {[N] \choose N-1} = P([N])$ we associate a map

$$R_K\colon \ \mathcal{U}_{\overrightarrow{P}(K)} \longrightarrow \mathcal{U}_{\overleftarrow{P}(K)}$$

The (N-1)-simplex equation

$$R_{\overrightarrow{P}([N])} = R_{\overleftarrow{P}([N])} \tag{3.1}$$

may then be regarded as a realization of B(N, N - 2). The expressions on both sides are compositions of maps R_K , $K \in P([N])$, applied on the left-hand side in lexicographic, on the right-hand side in reverse lexicographic order. Writing $\overrightarrow{P}([N]) = (K_1, \ldots, K_N)$, we have $R_{\overrightarrow{P}([N])} = R_{K_N} \cdots R_{K_1}$. Hence, as a composition, $R_{\overrightarrow{P}([N])}$ is actually in reverse lexicographic order, but the maps are applied in lexicographic order. We have to add the following rules in order for (3.1) to make sense.

- (1) Both sides of (3.1) act on \mathcal{U}_{α} and map to \mathcal{U}_{ω} , where α (ω) is $\binom{[N]}{N-2}$ in lexicographic (reverse lexicographic) order.
- (2) Each of the maps R_K acts at consecutive positions in the respective multiple Cartesian product of spaces.
- (3) If $J, J' \in {[N] \choose N-2}$ are such that they do not both belong to P(K) for any $K \in {[N] \choose N-1}$, then $\dots \times \mathcal{U}_I \times \mathcal{U}_{I'} \times \dots \sim \dots \times \mathcal{U}_{I'} \times \mathcal{U}_I \times \dots$

imposes an equivalence relation on Cartesian products.

Starting with \mathcal{U}_{α} , it may be necessary to use the third rule to arrange that R_{K_1} , respectively R_{K_N} , can be applied, which means that the sets associated with elements of $P(K_1)$, respectively $P(K_N)$, have to be in lexicographic order and at neighboring positions in the respective multiple Cartesian product of sets. After an application of some R_K , it may again be necessary to use the third rule to arrange a further application of a map $R_{K'}$, or to achieve the final reverse lexicographic order \mathcal{U}_{ω} . That this works is a consequence of the underlying structure of higher Bruhat orders [13, 42, 43].

We have to stress that (3.1) is *not* the form in which simplex equations usually appear in the literature, see [13] for the relation and references.

With each $K \in {\binom{[N]}{N-1}}$, now we associate a map

$$T_K\colon \ \mathcal{U}_{\overrightarrow{P}_{\mathrm{o}}(K)} \longrightarrow \mathcal{U}_{\overleftarrow{P}_{\mathrm{e}}(K)}$$

Writing $K = (k_1, ..., k_{N-1})$, with $k_i < k_{i+1}$, i = 1, ..., N-2, we have

$$\mathcal{U}_{\overrightarrow{P_{o}}(K)} = \mathcal{U}_{K \setminus \{k_{N-1}\}} \times \mathcal{U}_{K \setminus \{k_{N-3}\}} \times \cdots \times \mathcal{U}_{K \setminus \{k_{1+(N \mod 2)}\}},$$
$$\mathcal{U}_{\overleftarrow{P_{e}}(K)} = \mathcal{U}_{K \setminus \{k_{2-(N \mod 2)}\}} \times \cdots \times \mathcal{U}_{K \setminus \{k_{N-4}\}} \times \mathcal{U}_{K \setminus \{k_{N-2}\}}.$$

The N-gon equation

$$T_{\overrightarrow{P}_{o}([N])} = T_{\overleftarrow{P}_{e}([N])}$$
(3.2)

may be regarded as a realization of T(N, N-2). It is well defined if we require the following rules.

- (1) Let α (ω) be again $\binom{[N]}{N-2}$ in lexicographic (reverse lexicographic) order, and let $\alpha^{(b)}$ and $\omega^{(b)}$ be the corresponding blue parts. Both sides of (3.2) act on $\mathcal{U}_{\alpha^{(b)}}$ and map to $\mathcal{U}_{\omega^{(b)}}$.
- (2) Each of the maps T_K acts at consecutive positions in the respective multiple Cartesian product of sets.
- (3) If $J, J' \in {[N] \choose N-2}$ are such that they do not both belong to P(K) for any $K \in {[N] \choose N-1}$, then $\dots \times \mathcal{U}_J \times \mathcal{U}_{J'} \times \dots \sim \dots \times \mathcal{U}_{J'} \times \mathcal{U}_J \times \dots$.

As in the case of simplex equations, to apply or work out a polygon equation, we have to check at each step whether a map T_K can be applied directly or whether we first have to use the third rule above to achieve a reordering of the respective multiple Cartesian product. In any case, we have to keep track of the numbering of the sets, even if they are identical as sets.

It is, therefore, convenient to realize the above equivalence relation by introducing explicitly transposition maps (sometimes called flip or switch maps) in the equations, at the price of ending up with a form of the equation that looks more complicated and apparently lost its universal structure, but it is often better suited for applications. Instead of keeping track of the numbering of sets, we then have to keep track of the first position on which a map acts in a multiple Cartesian product. This has been done in [13]. For several polygon equations, we will recall the resulting form in Section 5. In this form, we can best deal with the case of prime interest, where all the sets U_J are the same and there is only a single map T.

If N is odd, (3.2) can be written as

$$T_{\hat{1}}T_{\hat{3}}\cdots T_{\widehat{N-2}}T_{\widehat{N}} = T_{\widehat{N-1}}T_{\widehat{N-3}}\cdots T_{\hat{2}},$$
(3.3)

where $\hat{k} := [N] \setminus \{k\}$ (complementary index notation).

If N is even, (3.2) can be correspondingly expressed as

$$T_{\hat{2}}T_{\hat{4}}\cdots T_{\widehat{N-2}}T_{\widehat{N}} = T_{\widehat{N-1}}T_{\widehat{N-3}}\cdots T_{\hat{1}}.$$
(3.4)

With each $K \in {[N] \choose N-1}$, N > 2, we also associate a map

$$\tilde{T}_K \colon \mathcal{U}_{\overrightarrow{P}_{\mathbf{e}}(K)} \longrightarrow \mathcal{U}_{\overleftarrow{P}_{\mathbf{o}}(K)}.$$

The dual N-gon equation

$$\tilde{T}_{\overrightarrow{P}_{e}([N])} = \tilde{T}_{\overleftarrow{P}_{o}([N])}$$
(3.5)

may be regarded as a realization of the dual Tamari order $\tilde{T}(N, N-2)$. Both sides act on $\alpha^{(r)}$, which is equal to $\omega^{(b)}$ totally reversed, and map to $\omega^{(r)}$, which is equal to $\alpha^{(b)}$ totally reversed.

For odd N, (3.5) is

$$\tilde{T}_{\hat{2}}\tilde{T}_{\hat{4}}\cdots\tilde{T}_{\widehat{N-1}}=\tilde{T}_{\widehat{N}}\tilde{T}_{\widehat{N-2}}\cdots\tilde{T}_{\hat{3}}\tilde{T}_{\hat{1}},$$

which is (3.3) reversed. For even N, we have

$$\tilde{T}_{\hat{1}}\tilde{T}_{\hat{3}}\cdots\tilde{T}_{\widehat{N-1}}=\tilde{T}_{\widehat{N}}\tilde{T}_{\widehat{N-2}}\cdots\tilde{T}_{\hat{2}},$$
(3.6)

which is (3.4) reversed.

Simplex equations, and also (dual) polygon equations, are interrelated by a kind of integrability feature, which crucially distinguishes them from similar equations. We refer to [13] for the general structure, but in Section 5 we elaborate this feature for some examples of polygon equations.

Remark 3.1. The dual (N+2)-gon equation is the N-cocycle condition in [57, 58]. In [57], the reader finds an explanation in which sense these equations can be regarded as "cocycles".

4 Reductions of polygon equations

For any fixed $k \in [N+1]$, there is a projection $\operatorname{pr}_k \colon A(N+1, n+1) \to A(N, n)$, obtained by restricting $\rho \in A(N+1, n+1)$ to the subsequence consisting only of elements $K \in {[N+1] \choose n+1}$ with $k \in K$. The set of all these subsequences is in bijection with A(N, n), simply by deleting k in each K and an obvious renumbering. Moreover, the projection is compatible with the equivalence relation \sim and induces a projection $B(N+1, n+1) \to B(N, n)$. See [13, Remark 2.5].

But only for $k \in \{1, N+1\}$, the projection pr_k is compatible with the 3-color decomposition, see [13, Remark 2.17]. If k = 1, this yields projections $T(N+1, n+1) \to T(N, n)$ and $\tilde{T}(N+1, n+1) \to \tilde{T}(N, n)$. For k = N+1, we have projections $T(N+1, n+1) \to \tilde{T}(N, n)$ and $\tilde{T}(N+1, n+1) \to T(N, n)$.

In particular, there are thus projections $T(N+1, N-1) \to T(N, N-2)$ and $T(N+1, N-1) \to \tilde{T}(N, N-2)$, which then induce reductions of the (N+1)-gon to the N-gon equation, respectively the dual N-gon equation. In the same way, the projections $\tilde{T}(N+1, N-1) \to \tilde{T}(N, N-2)$ and $\tilde{T}(N+1, N-1) \to T(N, N-2)$ induce reductions of the dual (N+1)-gon to the dual N-gon equation, respectively the N-gon equation. Since we realize Tamari orders by sets and maps between them, we have to make sure, however, that this procedure indeed leads to a realization of the (dual) N-gon equation. This will be made precise in the following subsections. Throughout we assume N > 2.

By a *degenerate* map (e.g., a solution of a polygon equation), we mean a map whose values do not depend on (at least) one of its arguments.

4.1 Reductions induced by pr_1

The reduction of the (N + 1)-gon equation induced by pr_1 is essentially obtained by dropping the map T_K with $1 \notin K$. But we also have to arrange that the remaining maps T_K with $1 \in K$ are reduced to maps between (products of) spaces U_J with $1 \in J \in P(K)$.

Theorem 4.1. Let $N \in \mathbb{N}$ be odd, N > 2.

- (1) Let T_K , $K \in {\binom{[N+1]}{N}}$, be maps solving the (N+1)-gon equation. For K with $1 \in K$, let T_K not depend on the last component of its domain. Let $T_{K'}$, $K' = K \setminus \{1\}$, be obtained from T_K by excluding $\mathcal{U}_{K'}$ from its domain. Then $\{T_{K'}|K' \in {\binom{\{2,\dots,N+1\}}{N-1}}\}$ solve the N-gon equation.
- (2) Each solution of the N-gon equation can be extended to a degenerate solution of the (N+1)-gon equation.

Proof. (1) If N is odd, then $K' = K \setminus \{1\}$ is the last element of $\overrightarrow{P}_{o}(K)$. If

$$T_K \colon \mathcal{U}_{\overrightarrow{P}_{0}(K)} \times \mathcal{U}_{K \setminus \{1\}} \to \mathcal{U}_{\overleftarrow{P}_{e}(K)}$$

does not depend on the last component of its domain, it induces a map

$$T_{K'}: \mathcal{U}_{\overrightarrow{P}_{0}(K')} \to \mathcal{U}_{\overleftarrow{P}_{e}(K')},$$

with $\mathcal{U}_J := \mathcal{U}_{\{1\}\cup J}$ for $J \in P(K')$. Since $T_{\hat{1}} = T_{\{2,3,\dots,N+1\}} : \mathcal{U}_{\overrightarrow{P}_o(\{2,3,\dots,N+1\})} \to \mathcal{U}_{\overrightarrow{P}_e(\{2,3,\dots,N+1\})}$ maps to spaces that are, as a consequence of our assumption, disregarded by all other maps, it can be dropped from the (N+1)-gon equation

$$T_{\hat{2}}T_{\hat{4}}\cdots T_{\widehat{N-1}}T_{\widehat{N+1}}=T_{\widehat{N}}T_{\widehat{N-2}}\cdots T_{\hat{1}},$$

which thus reduces to

$$T_{\hat{2}}T_{\hat{4}}\cdots T_{\widehat{N-1}}T_{\widehat{N+1}}=T_{\widehat{N}}T_{\widehat{N-2}}\cdots T_{\hat{3}},$$

where we can now regard the indices as complementary in $[N + 1] \setminus \{1\}$. By a shift in the numbering, this is turned into the standard form of the odd N-gon equation,

$$T_{\hat{1}}T_{\hat{3}}\cdots T_{\widehat{N-2}}T_{\widehat{N}}=T_{\widehat{N-1}}T_{\widehat{N-3}}\cdots T_{\hat{2}},$$

where the indices are complementary in [N].

(2) Applying a shift in the numbering of maps, constituting a solution of the N-gon equation, it is given by $\{T_{K'}|K' \in \binom{\{2,\ldots,N+1\}}{N-1}\}$. Associating $K = \{1\} \cup K'$ with K', we trivially extend the map $T_{K'}$ to a map T_K of the form given above, by introducing a set $\mathcal{U}_{K\setminus\{1\}}$. Choosing furthermore a map $T_{\{2,3,\ldots,N+1\}}: \mathcal{U}_{\overrightarrow{P}_o(\{2,3,\ldots,N+1\})} \to \mathcal{U}_{\overrightarrow{P}_e(\{2,3,\ldots,N+1\})}$, with sets \mathcal{U}_J , $J \in P(\{2,3,\ldots,N+1\})$, we obtain a solution of the (N+1)-gon equation. This essentially reverses the steps taken in the proof of (1).

Theorem 4.2. Let $N \in \mathbb{N}$ be even, N > 2. Let T_K , $K \in \binom{[N+1]}{N}$, be maps solving the (N+1)gon equation. For K with $1 \in K$ let $T_{K'}$, $K' = K \setminus \{1\}$, be obtained from T_K by deleting the
first component of its codomain. Then $\{T_{K'} | K' \in \binom{\{2, \dots, N+1\}}{N-1}\}$ solve the N-gon equation.

Proof. Let N be even and K such that $1 \in K$. The packet P(K) has N elements and its last member in the lexicographic order is $K \setminus \{1\}$. Hence, $K \setminus \{1\}$ is the first element of $\overline{P}_{e}(K)$. Then

$$T_K\colon \mathcal{U}_{\overrightarrow{P}_{o}(K)} \to \mathcal{U}_{K\setminus\{1\}} \times \mathcal{U}_{\overleftarrow{P}'_{e}(K)},$$

where $\overleftarrow{P}'_{e}(K)$ is $\overleftarrow{P}_{e}(K)$ without the element $K' := K \setminus \{1\}$. By disregarding the first component of its codomain, each T_{K} , $1 \in K$, induces a map

$$T_{K'}\colon \mathcal{U}_{\overrightarrow{P}_{o}(K')} \to \mathcal{U}_{\overleftarrow{P}_{e}(K')}, \qquad K' \in \binom{\{2, \dots, N+1\}}{N-1},$$

where, for $J \in P(K')$, we set again $\mathcal{U}_J := \mathcal{U}_{\{1\}\cup J}$. Since the domain of the remaining map $T_{\hat{1}}$ only involves spaces that are excluded from the range of all other maps T_K , it splits off from the (N + 1)-gon equation

$$T_{\hat{1}}T_{\hat{3}}\cdots T_{\widehat{N-1}}T_{\widehat{N+1}}=T_{\widehat{N}}T_{\widehat{N-2}}\cdots T_{\hat{2}},$$

which thus reduces to

$$T_{\hat{3}}\cdots T_{\widehat{N-1}}T_{\widehat{N+1}}=T_{\widehat{N}}T_{\widehat{N-2}}\cdots T_{\hat{2}},$$

where the complementary indices now refer to $[N+1] \setminus \{1\} = \{2, \ldots, N+1\}$. By renaming $T_{\hat{k}}$ to $T_{\widehat{k-1}}$ in the last equation, it reads

$$T_{\hat{2}}\cdots T_{\widehat{N-2}}T_{\widehat{N}}=T_{\widehat{N-1}}T_{\widehat{N-3}}\cdots T_{\hat{1}},$$

where now the indices are complementary in [N], so that we have the standard form of the (even) N-gon equation.

Remark 4.3. The last result means that, if $\{T_K | K \in {\binom{[N+1]}{N}}\}$ is a solution of an odd (N+1)-gon equation, then each map T_K with $1 \in K$ has the form

$$T_K = S_{K'} \times T_{K'}, \qquad K' := K \setminus \{1\},$$

with a map $S_{K'}: \mathcal{U}_{\overrightarrow{P}_{o}(K)} \to \mathcal{U}_{K'}$, and $\{T_{K'} | K' \in \binom{\{2,3,\dots,N+1\}}{N-1}\}$ solve the even N-gon equation. Each solution of the odd (N+1)-gon equation is thus an extension, of the above form, of a solution of the (even) N-gon equation.

Example 4.4. The pentagon equation $T_{\hat{1}}T_{\hat{3}}T_{\hat{5}} = T_{\hat{4}}T_{\hat{2}}$ involves the maps $T_{\hat{1}}: \mathcal{U}_{234} \times \mathcal{U}_{245} \rightarrow \mathcal{U}_{345} \times \mathcal{U}_{235}, T_{\hat{2}}: \mathcal{U}_{134} \times \mathcal{U}_{145} \rightarrow \mathcal{U}_{345} \times \mathcal{U}_{135}, T_{\hat{3}}: \mathcal{U}_{124} \times \mathcal{U}_{145} \rightarrow \mathcal{U}_{245} \times \mathcal{U}_{125}, T_{\hat{4}}: \mathcal{U}_{123} \times \mathcal{U}_{135} \rightarrow \mathcal{U}_{235} \times \mathcal{U}_{125}, T_{\hat{5}}: \mathcal{U}_{123} \times \mathcal{U}_{134} \rightarrow \mathcal{U}_{234} \times \mathcal{U}_{124}.$ We thus have

$$T_{\hat{4}}T_{\hat{2}}: \mathcal{U}_{123} \times \mathcal{U}_{134} \times \mathcal{U}_{145} \to (\mathcal{U}_{345} \times \mathcal{U}_{235}) \times \mathcal{U}_{125}$$

and

$$T_{3}T_{5}: \mathcal{U}_{123} \times \mathcal{U}_{134} \times \mathcal{U}_{145} \to (\mathcal{U}_{234} \times \mathcal{U}_{245}) \times \mathcal{U}_{125}$$

Recalling that $T_{\hat{1}}$ does not involve any of the spaces \mathcal{U}_{1ij} , $1 < i < j \leq 5$, the pentagon equation implies that the maps, obtained from $\{T_{\hat{k}}\}$ by deleting the first component of their codomain, satisfy the tetragon equation (with indices shifted by 1).

Let us now turn to the dual (N+1)-gon equation and consider the subset of maps $\tilde{T}_K \colon \mathcal{U}_{\overrightarrow{P}_e(K)} \to \mathcal{U}_{\overrightarrow{P}_o(K)}$ with $1 \in K$.

Theorem 4.5. Let $N \in \mathbb{N}$ be even, N > 2.

- (1) Let \tilde{T}_K , $K \in {\binom{[N+1]}{N}}$, be maps solving the dual (N+1)-gon equation. For K with $1 \in K$, let \tilde{T}_K not depend on the last component of its domain. Let $\tilde{T}_{K'}$, $K' = K \setminus \{1\}$, be obtained from \tilde{T}_K by excluding the last component of its domain. Then $\{\tilde{T}_{K'}|K' \in {\binom{\{2,\dots,N+1\}}{N-1}}\}$ solve the dual N-gon equation.
- (2) Each solution of the dual N-gon equation can be extended to a degenerate solution of the dual (N + 1)-gon equation.

Proof. (1) If N is even, then $K' = K \setminus \{1\}$ is the last element of $\overrightarrow{P}_{e}(K)$, so that

$$\tilde{T}_K: \ \mathcal{U}_{\overrightarrow{P}'_{\mathrm{e}}(K)} \times \mathcal{U}_{K \setminus \{1\}} \to \mathcal{U}_{\overleftarrow{P}_{\mathrm{o}}(K)},$$

where $\overrightarrow{P}'_{\rm e}(K)$ is $\overrightarrow{P}_{\rm e}(K)$ without the last element. If \widetilde{T}_K does not depend on the last component of its domain, it induces a map

$$\widetilde{T}_{K'}: \ \mathcal{U}_{\overrightarrow{P}_{e}(K')} \to \mathcal{U}_{\overleftarrow{P}_{o}(K')}, \qquad K' \in \binom{\{2, \dots, N+1\}}{N-1},$$

where we set $\mathcal{U}_J := \mathcal{U}_{\{1\}\cup J}$. With the same argument as in the proof of Theorem 4.1, the dual (N+1)-gon equation

$$\tilde{T}_{\hat{2}}\tilde{T}_{\hat{4}}\cdots\tilde{T}_{\widehat{N}}=\tilde{T}_{\widehat{N+1}}\tilde{T}_{\widehat{N-1}}\cdots\tilde{T}_{\hat{3}}\tilde{T}_{\hat{1}}$$

reduces to

$$\tilde{T}_{\hat{2}}\tilde{T}_{\hat{4}}\cdots\tilde{T}_{\widehat{N}}=\tilde{T}_{\widehat{N+1}}\tilde{T}_{\widehat{N-1}}\cdots\tilde{T}_{\hat{3}},$$

where now the complementary indices refer to $[N+1] \setminus \{1\}$. A shift in the numbering achieves the standard form (3.6) of the dual even N-gon equation.

(2) The proof is analogous to that of part (2) of Theorem 4.1.

Theorem 4.6. Let $N \in \mathbb{N}$ be odd, N > 2. Let \tilde{T}_K , $K \in {\binom{[N+1]}{N}}$, be maps solving the dual (N+1)gon equation. For K with $1 \in K$ let $\tilde{T}_{K'}$, $K' = K \setminus \{1\}$, be obtained from \tilde{T}_K by excluding the
first component of its codomain. Then $\{\tilde{T}_{K'}|K' \in {\binom{\{2,\dots,N+1\}}{N-1}}\}$ solve the dual N-gon equation.

Proof. If N is odd and $1 \in K$, then K' is the first element of $\overleftarrow{P}_{o}(K)$, so that

$$\widetilde{T}_K: \ \mathcal{U}_{\overrightarrow{P}_{\mathrm{e}}(K)} \to \mathcal{U}_{K\setminus\{1\}} \times \mathcal{U}_{\overleftarrow{P}_{\mathrm{o}}(K)},$$

where $\overleftarrow{P}'_{o}(K)$ is $\overleftarrow{P}_{o}(K)$ without the first element. Now each \widetilde{T}_{K} , $1 \in K$, induces a map

$$\widetilde{T}_{K'}: \ \mathcal{U}_{\overrightarrow{P}_{e}(K')} \to \mathcal{U}_{\overleftarrow{P}_{o}(K')}, \qquad K' \in \binom{\{2, \dots, N+1\}}{N-1}.$$

The dual (N+1)-gon equation then reduces to the dual odd N-gon equation, using an argument as in the proofs of the preceding theorems.

4.2 Reductions induced by pr_{N+1}

The projection $T(N + 1, N - 1) \rightarrow \tilde{T}(N, N - 2)$ induces a reduction of the (N + 1)-gon to the dual N-gon equation. It is essentially obtained by dropping the map T_K with $N + 1 \notin K$. But we have to arrange that the remaining maps T_K , $N + 1 \in K$, are reduced to maps between (products of) spaces U_J with $N + 1 \in J \in P(K)$.

Theorem 4.7. Let $N \in \mathbb{N}$, N > 2.

- (1) Let T_K , $K \in {\binom{[N+1]}{N}}$, be maps solving the (N+1)-gon equation. For K with $N+1 \in K$, let T_K not depend on the first component of its domain. Set $K' = K \setminus \{N+1\}$ and let $\tilde{T}_{K'}$ be given by T_K with the first component of its domain excluded. Then $\{\tilde{T}_{K'}|K' \in {\binom{[N]}{N-1}}\}$ solve the dual N-gon equation.
- (2) Each solution of the dual N-gon equation can be extended to a degenerate solution of the (N + 1)-gon equation.

Proof. (1) Let K be such that $N+1 \in K$. Then $K' := K \setminus \{N+1\}$ is the first element of P(K) in lexicographic order, and thus also of $P_0(K)$. Hence, we have maps

$$T_K: \ \mathcal{U}_{K\setminus\{N+1\}} \times \mathcal{U}_{\overrightarrow{P}_o(K)} \to \mathcal{U}_{\overleftarrow{P}_e(K)}, \qquad K \in \binom{[N+1]}{N}, \qquad N+1 \in K,$$

where $P'_{o}(K)$ is $P_{o}(K)$ without its first element (in lexicographic order). If these maps do not depend on the first component of their domain, they project to maps

$$\tilde{T}_{K'}: \ \mathcal{U}_{\overrightarrow{P}_{\mathrm{e}}(K')} \to \mathcal{U}_{\overleftarrow{P}_{\mathrm{o}}(K')}, \qquad K' \in \binom{[N]}{N-1},$$

where we set $\mathcal{U}_J := \mathcal{U}_{J \cup \{N+1\}}$ for $J \in P(K')$. Under this assumption,

$$T_{\widehat{N+1}} \colon \ \mathcal{U}_{\overrightarrow{P}_{o}(\{1,2,\dots,N\})} \to \mathcal{U}_{\overleftarrow{P}_{e}(\{1,2,\dots,N\})}$$

maps to spaces that are disregarded by all other T_K appearing in the (N+1)-gon equation.

For odd N, the (N + 1)-gon equation

$$T_{\hat{2}}T_{\hat{4}}\cdots T_{\widehat{N-1}}T_{\widehat{N+1}}=T_{\widehat{N}}T_{\widehat{N-2}}\cdots T_{\hat{1}}$$

thus reduces to

$$\tilde{T}_{\hat{2}}\tilde{T}_{\hat{4}}\cdots\tilde{T}_{\widehat{N-1}}=\tilde{T}_{\widehat{N}}\tilde{T}_{\widehat{N-2}}\cdots\tilde{T}_{\hat{3}}\tilde{T}_{\hat{1}},$$

where now the indices are complementary in $[N + 1] \setminus \{N + 1\} = [N]$. This is the dual N-gon equation for odd N.

For even N, the (N + 1)-gon equation

$$T_{\hat{1}}T_{\hat{3}}\cdots T_{\widehat{N-1}}T_{\widehat{N+1}} = T_{\widehat{N}}T_{\widehat{N-2}}\cdots T_{\hat{2}}$$

reduces to

$$\tilde{T}_{\hat{1}}\tilde{T}_{\hat{3}}\cdots\tilde{T}_{\widehat{N-1}}=\tilde{T}_{\widehat{N}}\tilde{T}_{\widehat{N-2}}\cdots\tilde{T}_{\hat{2}}$$

where again indices are now complementary in [N]. This is the dual N-gon equation for even N.

(2) The proof is analogous to that in preceding theorems.

Let us now turn to the corresponding reduction of the dual (N+1)-gon equation and consider the maps $\tilde{T}_K: \mathcal{U}_{\overrightarrow{P}_o(K)} \to \mathcal{U}_{\overrightarrow{P}_o(K)}$ with $N+1 \in K$.

Theorem 4.8. Let $N \in \mathbb{N}$, N > 2. Let \tilde{T}_K , $K \in {\binom{[N+1]}{N}}$, be maps solving the dual (N + 1)-gon equation. For K with $N + 1 \in K$, set $K' = K \setminus \{N + 1\}$ and let $T_{K'}$ be given by \tilde{T}_K with the last component of its codomain excluded. Then $\{T_{K'} | K' \in {\binom{[N]}{N-1}}\}$ solve the N-gon equation.

Proof. Since $K' = K \setminus \{N+1\}$ is the first element of $\overrightarrow{P}(K)$, and thus the last element of $\overleftarrow{P}_{o}(K)$,

$$\tilde{T}_K: \ \mathcal{U}_{\overrightarrow{P}_{\mathrm{e}}(K)} \to \mathcal{U}_{\overleftarrow{P}_{\mathrm{o}}(K)} \times \mathcal{U}_{K \setminus \{N+1\}},$$

where $\overleftarrow{P}'_{o}(K)$ is $\overleftarrow{P}_{o}(K)$ without the last element. Hence, \widetilde{T}_{K} induces a map

$$T_{K'}: \mathcal{U}_{\overrightarrow{P}_{o}(K')} \to \mathcal{U}_{\overleftarrow{P}_{e}(K')},$$

where we set again $\mathcal{U}_J := \mathcal{U}_{J \cup \{N+1\}}$ for $J \in P(K')$. With the kind of argument used in the proofs of preceding theorems, the dual (N+1)-gon equation reduces to the N-gon equation.

Remark 4.9. The last result means that, if $\{\tilde{T}_K | K \in {\binom{[N+1]}{N}}\}$ is a solution of a dual (N+1)-gon equation, then each map \tilde{T}_K with $N+1 \in K$ has the form

$$\tilde{T}_K = T_{K'} \times S_{K'}, \qquad K' := K \setminus \{N+1\},\$$

with a map $S_{K'}: \mathcal{U}_{\overrightarrow{P}_{e}(K)} \to \mathcal{U}_{K'}$, and $\{T_{K'} | K' \in {[N] \choose N-1}\}$ solve the N-gon equation. Each solution of the dual (N+1)-gon equation is thus an extension, of the above form, of a solution of the N-gon equation.

5 Examples of polygon equations

In this section, we elaborate polygon equations up to the 8-gon equation.

5.1 Trigon equation

This is the equation

$$T_{23}T_{12} = T_{13} \tag{5.1}$$

for maps $T_{ij}: \mathcal{U}_i \to \mathcal{U}_j$, i < j. On the left-hand side of (5.1), we mean the composition of two maps. If the sets are the same, $\mathcal{U}_i = \mathcal{U}$, i = 1, 2, 3, and if there is only a single map T, then the trigon equation means that T is idempotent.



Figure 1. From local trigon equations to the tetragon equation. Here, and correspondingly in following figures, we suppress the parameters of the maps L.

5.2 Tetragon equation

For i, j = 1, 2, 3, 4, i < j, let a map $L_{ij}: \mathcal{U}_i \to \mathcal{U}_j$ carry a parameter from a set \mathcal{U}_{ij} . Let us further assume that each of the *local trigon equations*

$$L_{jk}(u_{jk})L_{ij}(u_{ij}) = L_{ik}(T_{ijk}(u_{ij}, u_{jk})), \quad i < j < k,$$

uniquely determines a map

 $T_{ijk}: \mathcal{U}_{ij} \times \mathcal{U}_{jk} \longrightarrow \mathcal{U}_{ik}.$

Using associativity of compositions, there is a consistency condition, see Figure 1,⁶ which requires that the maps T_{ijk} have to satisfy the *tetragon equation*

 $T_{134}T_{123} = T_{124}T_{234}.$

Both sides of this equation act on the lexicographically ordered Cartesian product $\mathcal{U}_{12} \times \mathcal{U}_{23} \times \mathcal{U}_{34}$ and map to \mathcal{U}_{14} .

Let us introduce a boldface "position index" that indicates the first of two neighboring sets on which the respective map acts in a Cartesian product of more than two sets,

 $T_{134}T_{123,1} = T_{124}T_{234,2}.$

These additional indices are redundant, as long as we keep the combinatorial indices and keep track of the numbered sets.

Using complementary index notation, where \hat{k} stands for the complement of k in $\{1, 2, 3, 4\}$, the tetragon equation reads $T_{\hat{2}}T_{\hat{4},1} = T_{\hat{3}}T_{\hat{1},2}$.

Writing

$$T_{ijk}(a_{ij}, a_{jk}) =: a_{ij} \bullet_{ijk} a_{jk},$$

with $a_{ij} \in \mathcal{U}_{ij}$, the equation takes the form

 $(a_{12} \bullet_{123} a_{23}) \bullet_{134} a_{34} = a_{12} \bullet_{124} (a_{23} \bullet_{234} a_{34}),$

which is a mixed associativity condition for the (in general different) binary operations \bullet_{123} , \bullet_{124} , \bullet_{134} and \bullet_{234} .⁷

In the simplest case, where all the basic sets are equal and we are dealing with a single map T, we may drop the combinatorial indices, but retain the boldface "position" indices. The tetragon equation is then $TT_1 = TT_2$. Writing $T(a, b) = a \cdot b$, it becomes the associativity relation for the binary operation \cdot .

⁶Such a diagram already appeared in the 1965 PhD Thesis of James Wirth, see [68], where the maps are, however, of a different nature, so that the step to higher polygon equations does not work.

⁷Examples of such associativity relations for different binary operations are provided, for example, by nonsymmetric Poisson algebras (in a setting of vector spaces, with \times replaced by the corresponding tensor product) [44].



Figure 2. From local tetragon equations to the pentagon equation. Here \sim stands for an equivalence, which corresponds to an application of a transposition map \mathcal{P} . Essentially, the left diagram is the pentagonal *Tamari lattice*, in its original form, as displayed in the second figure. Here the action of a map T corresponds to a right associativity map $(a \cdot b) \cdot c \mapsto a \cdot (b \cdot c)$. If this is an invertible map in a category with a binary operation \cdot , the pentagon relation means that it is a *coherent associativity isomorphism* in the sense of [40].

5.3 Pentagon equation

For i, j, k = 1, ..., 5, i < j < k, let a map

 $L_{ijk}: \mathcal{U}_{ij} \times \mathcal{U}_{jk} \longrightarrow \mathcal{U}_{ik}$

depend on a parameter from a set \mathcal{U}_{ijk} . Let us assume that each of the local tetragon equations

 $L_{ikl}(u_{ikl})L_{ijk,1}(u_{ijk}) = L_{ijl}(v_{ijl})L_{jkl,2}(v_{jkl}), \qquad 1 \le i < j < k < l \le 5,$

uniquely determines a map

 $T_{ijkl}: \mathcal{U}_{ijk} \times \mathcal{U}_{ikl} \to \mathcal{U}_{jkl} \times \mathcal{U}_{ijl}$

via $(u_{ijk}, u_{ikl}) \mapsto (v_{jkl}, v_{ijl})$. Then it follows that the maps T_{ijkl} , $1 \le i < j < k < l \le 5$, have to satisfy the pentagon equation, see Figure 2.

Using complementary index notation, the *pentagon equation* is

$$T_{\hat{1}}T_{\hat{3}}T_{\hat{5}} = T_{\hat{4}}T_{\hat{2}}$$

Both sides of this equation act on $\mathcal{U}_{123} \times \mathcal{U}_{134} \times \mathcal{U}_{145}$ and map to $\mathcal{U}_{345} \times \mathcal{U}_{235} \times \mathcal{U}_{125}$.

Representing the equivalence relation \sim in the diagram in Figure 2 by a transposition map, $\mathcal{P}(a,b) = (b,a)$, and reading off on which neighboring positions in a multiple Cartesian product a map acts, we get

$$T_{\hat{1},\mathbf{1}}T_{\hat{3},\mathbf{2}}T_{\hat{5},\mathbf{1}} = T_{\hat{4},\mathbf{2}}\mathcal{P}_{\mathbf{1}}T_{\hat{2},\mathbf{2}}.$$

In case of identical basic sets, then renamed to \mathcal{U} , and a single map T, the last equation takes the form

$$T_1 T_2 T_1 = T_2 \mathcal{P}_1 T_2, \tag{5.2}$$

where the combinatorial indices have been dropped. Now all the information needed is provided by the "position" indices. The latter is our abbreviation of the following more familiar form of the pentagon equation

$$T_{12}T_{23}T_{12} = T_{23}\mathcal{P}_{12}T_{23}.$$

Writing

$$T(a,b) = (a * b, a \cdot b), \tag{5.3}$$

the last restricted form of the pentagon equation is equivalent to the conditions (cf. [30, 31])

$$(a * b) * ((a \cdot b) * c) = b * c, \qquad (a * b) \cdot ((a \cdot b) * c) = a * (b \cdot c), (a \cdot b) \cdot c = a \cdot (b \cdot c),$$
(5.4)

for all $a, b, c \in \mathcal{U}$. Set-theoretic solutions have been obtained in [1, 5, 6, 7, 22, 25, 26, 27, 28, 31, 32, 45, 46, 70].

Example 5.1.

- (1) If a * b = b for all $a, b \in \mathcal{U}$, (5.4) reduces to the associativity condition for \cdot . If (\mathcal{U}, \cdot) is a group and if T is invertible, then this is the only solution [7, 31]. It underlies one of the Kac–Takesaki operators on a group (see, e.g., [62]). If \mathcal{U} is a subset of a group (G, \cdot) , not containing the identity element, there are more solutions [31].
- (2) If $a \cdot b = a$, the above system reduces to (a * b) * (a * c) = b * c. In a group, a solution is given by $a * b = a^{-1}b$. This underlies another Kac–Takesaki operator on a group (see, e.g., [5, 62]).

In terms of the composition $\hat{T} := T\mathcal{P}$, (5.2) takes the form

$$\hat{T}_{12}\hat{T}_{13}\hat{T}_{23} = \hat{T}_{23}\hat{T}_{12}.$$
(5.5)

Remark 5.2. If T solves (5.2), then $\check{T} := \mathcal{P}T$ satisfies the – relative to (5.5) – *reversed* pentagon equation

$$\check{T}_{12}\check{T}_{23} = \check{T}_{23}\check{T}_{13}\check{T}_{12}.$$
(5.6)

If an invertible map \hat{T} satisfies (5.5), then its inverse \hat{T}^{-1} satisfies the above reversed equation.

Remark 5.3. If \hat{T} is involutive, then it satisfies both, (5.5) and (5.6). Such solutions have been explored in [7].

5.3.1 An example related to incidence geometry

Let \mathcal{V} be a vector space and $L(a): \mathcal{V} \times \mathcal{V} \to \mathcal{V}$, $a \in \mathcal{U}$, be such that the "local" tetragon equation $L(b)L(a)_1 = L(b')L(a')_2$ determines a unique map $T: \mathcal{U} \times \mathcal{U} \to \mathcal{U} \times \mathcal{U}$ via $(a, b) \mapsto (a', b')$. Then this map satisfies the pentagon equation (5.2). Writing $x \circ_a y := L(a)(x, y)$, we can express the above equation as the parameter-dependent associativity condition $(x \circ_a y) \circ_b z = x \circ_{b'} (y \circ_{a'} z)$.

An example is given by $\mathcal{V} = \mathbb{R}^n$, $\mathcal{U} = (0, 1) \subset \mathbb{R}$, and $x \circ_a y := ax + (1 - a)y$. Then we obtain the solution

$$T(a,b) = \left(\frac{(1-a)b}{1-ab}, ab\right)$$
(5.7)



Figure 3. A $(6_2, 4_3)$ configuration.

of the pentagon equation (also see [26]). It also shows up as a map of "polarizations" resulting from the evolution of tree-shaped matrix KP solitons [14].

The above binary operation can be interpreted as the collinearity of three points $A, B, A \circ_a B$: $A \circ_a B = aA + (1-a)B$. The generalized associativity condition can then be viewed as a $(6_2, 4_3)$ configuration [21], which consists of 6 points and 4 lines, where each point is incident with exactly two lines, each line with exactly 3 points. See Figure 3. A Desargues configuration (10₃) consists of 10 points and 10 lines, each point (line) incident with 3 lines (points). It contains 5 configurations of type $(6_2, 4_3)$, which thus constitute a pentagon. Also see [15] for related considerations.

5.4 Hexagon equation

For i, j, k, l = 1, ..., 6, i < j < k < l, let a map

 $L_{ijkl}: \mathcal{U}_{ijk} \times \mathcal{U}_{ikl} \longrightarrow \mathcal{U}_{jkl} \times \mathcal{U}_{ijl}$

depend on a parameter from a set \mathcal{U}_{ijkl} . We assume that each of the local pentagon equations

$$L_{jklm}(u_{jklm})L_{ijlm}(u_{ijlm})L_{ijkl}(u_{ijkl}) = L_{ijkm}(v_{ijkm})L_{iklm}(v_{iklm}),$$

$$1 \le i \le j \le k \le l \le m \le 6.$$

uniquely determines a map

 $T_{ijklm}: \mathcal{U}_{ijkl} \times \mathcal{U}_{ijlm} \times \mathcal{U}_{jklm} \to \mathcal{U}_{iklm} \times \mathcal{U}_{ijkm}$

via $(u_{ijkl}, u_{ijlm}, u_{jklm}) \mapsto (v_{iklm}, v_{ijkm}).$

Then the maps T_{ijklm} , $1 \le i < j < k < l < m \le 6$, have to satisfy the hexagon equation. See Figure 4. Using complementary index notation, the *hexagon equation* reads

$$T_{\hat{2}}T_{\hat{4}}T_{\hat{6}} = T_{\hat{5}}T_{\hat{3}}T_{\hat{1}}.$$

Both sides act on $\mathcal{U}_{1234} \times \mathcal{U}_{1245} \times \mathcal{U}_{1256} \times \mathcal{U}_{2345} \times \mathcal{U}_{2356} \times \mathcal{U}_{3456}$ and map to $\mathcal{U}_{1456} \times \mathcal{U}_{1346} \times \mathcal{U}_{1236}$.

Employing transposition maps, according to the equivalences \sim appearing in Figure 4, and introducing position indices, the hexagon equation takes the form

$$T_{\hat{2},1}\mathcal{P}_{3}T_{\hat{4},2}T_{\hat{6},1}\mathcal{P}_{3} = T_{\hat{5},2}\mathcal{P}_{1}T_{\hat{3},2}T_{\hat{1},4}$$

If all basic sets are equal and we are dealing with a single map T, we may drop the combinatorial indices and only retain the (boldface) position indices:

$$T_1 \mathcal{P}_3 T_2 T_1 \mathcal{P}_3 = T_2 \mathcal{P}_1 T_2 T_4.$$
(5.8)



Figure 4. From local pentagon equations to the hexagon equation.

Remark 5.4. According to our knowledge, without introduction of an auxiliary structure as discussed in [13], the hexagon equation (5.8), and correspondingly all higher *even* polygon equations, cannot be rewritten without the explicit appearance of transpositions. This is in contrast to the case of the pentagon equation (see (5.5)) and higher *odd* polygon equations.

Expressing T in terms of two ternary operations,

$$T(a, b, c) = (\langle a, b, c \rangle, [a, b, c]), \tag{5.9}$$

the hexagon equation, acting on $(a, b, c, d, e, f) \in \mathcal{U}^6$, is equivalent to the following conditions,

$$\langle \langle a, b, d \rangle, \langle [a, b, d], c, e \rangle, f \rangle = \langle b, c, \langle d, e, f \rangle \rangle, [\langle a, b, d \rangle, \langle [a, b, d], c, e \rangle, f] = \langle a, [b, c, \langle d, e, f \rangle], [d, e, f] \rangle, [[a, b, d], c, e] = [a, [b, c, \langle d, e, f \rangle], [d, e, f]].$$

$$(5.10)$$

5.5 Heptagon equation

Let maps

$$L_{ijklm}: \mathcal{U}_{ijkl} \times \mathcal{U}_{ijlm} \times \mathcal{U}_{jklm} \to \mathcal{U}_{iklm} \times \mathcal{U}_{ijkm}$$

where $1 \le i < j < k < l < m \le 7$, be subject to local hexagon equations

$$L_{iklmn}(u_{iklmn})L_{ijkmn}(u_{ijkmn})L_{ijklm}(u_{ijklm})$$

= $L_{ijkln}(v_{ijkln})L_{ijlmn}(v_{ijlmn})L_{jklmn}(v_{jklmn}),$

where $1 \le i < j < k < l < m < n \le 7$. If these equations uniquely determine maps

 $T_{ijklmn}: \ \mathcal{U}_{ijklmn} \times \mathcal{U}_{ijkmn} \times \mathcal{U}_{iklmn} \to \mathcal{U}_{jklmn} \times \mathcal{U}_{ijlmn} \times \mathcal{U}_{ijklnn}$

via $(u_{ijklm}, u_{ijkmn}, u_{iklmn}) \mapsto (v_{jklmn}, v_{ijlmn}, v_{ijkln})$, elaborating

 $L_{14567}L_{13467}L_{12367}L_{13456}L_{12356}L_{12345} = L_{\widehat{23}}L_{\widehat{25}}L_{\widehat{45}}L_{\widehat{27}}L_{\widehat{47}}L_{\widehat{67}}$

in two different ways, using local hexagon equations, we find that the maps T_{ijklmn} have to satisfy the heptagon equation, which is

 $T_{\hat{1}}T_{\hat{3}}T_{\hat{5}}T_{\hat{7}} = T_{\hat{6}}T_{\hat{4}}T_{\hat{2}},$

in complementary index notation.

Introducing position indices and transposition maps, we can express it as

$$T_{\hat{1},1}T_{\hat{3},3}\mathcal{P}_{5}\mathcal{P}_{2}T_{\hat{5},3}T_{\hat{7},1}\mathcal{P}_{3} = \mathcal{P}_{3}T_{\hat{6},4}\mathcal{P}_{3}\mathcal{P}_{2}\mathcal{P}_{1}T_{\hat{4},3}\mathcal{P}_{2}\mathcal{P}_{3}T_{\hat{2},4}.$$

Both sides act on $\mathcal{U}_{12345} \times \mathcal{U}_{12356} \times \mathcal{U}_{12367} \times \mathcal{U}_{13456} \times \mathcal{U}_{13467} \times \mathcal{U}_{14567}$ and map to $\mathcal{U}_{34567} \times \mathcal{U}_{23567} \times \mathcal{U}_{23457} \times \mathcal{U}_{12567} \times \mathcal{U}_{12457} \times \mathcal{U}_{12347}$ (which can be read off from [13, Figure 18]).

If all basic sets are equal and there is only a single map T, all the information we need is in the position indices, so that the combinatorial indices can be dropped,

$$T_1T_3\mathcal{P}_5\mathcal{P}_2T_3T_1\mathcal{P}_3=\mathcal{P}_3T_4\mathcal{P}_3\mathcal{P}_2\mathcal{P}_1T_3\mathcal{P}_2\mathcal{P}_3T_4$$

Expressing T in terms of three ternary operations,⁸

$$T(a, b, c) = (\{a, b, c\}, \langle a, b, c \rangle, [a, b, c]),$$
(5.11)

the heptagon equation, evaluated on $(a, b, c, d, e, f) \in \mathcal{U}^6$, is equivalent to (5.10), supplemented by

$$\{\{a, b, d\}, \{[a, b, d], c, e\}, \{\langle a, b, d \rangle, \langle [a, b, d], c, e \rangle, f\}\} = \{d, e, f\}, \\ \langle \{a, b, d\}, \{[a, b, d], c, e\}, \{\langle a, b, d \rangle, \langle [a, b, d], c, e \rangle, f\}\rangle = \{b, c, \langle d, e, f \rangle\}, \\ [\{a, b, d\}, \{[a, b, d], c, e\}, \{\langle a, b, d \rangle, \langle [a, b, d], c, e \rangle, f\}] = \{a, [b, c, \langle d, e, f \rangle], [d, e, f]\}.$$
(5.12)

5.6 Octagon equation

We consider maps

 $T_{ijklmpq}: \ \mathcal{U}_{ijklmpq} \times \mathcal{U}_{ijklmpq} \times \mathcal{U}_{ijklmpq} \times \mathcal{U}_{ijklmpq} \rightarrow \mathcal{U}_{iklmpq} \times \mathcal{U}_{ijklmpq} \times \mathcal{U}_{ijklmpq}$

where i < j < k < l < m < p < q. The octagon equation is

$$T_{\hat{2}}T_{\hat{4}}T_{\hat{6}}T_{\hat{8}} = T_{\hat{7}}T_{\hat{5}}T_{\hat{3}}T_{\hat{1}}.$$

It arises as the consistency condition of a system of local heptagon equations.

Using position indices and transposition maps, it can be written as

$$T_{\hat{2},1}\mathcal{P}_{4}\mathcal{P}_{5}\mathcal{P}_{6}T_{\hat{4},3}\mathcal{P}_{6}\mathcal{P}_{5}\mathcal{P}_{2}T_{\hat{6},3}\mathcal{P}_{6}T_{\hat{8},1}\mathcal{P}_{4}\mathcal{P}_{5}\mathcal{P}_{6}\mathcal{P}_{3}$$
$$=\mathcal{P}_{3}T_{\hat{7},4}\mathcal{P}_{3}\mathcal{P}_{2}\mathcal{P}_{1}T_{\hat{5},3}\mathcal{P}_{6}\mathcal{P}_{2}\mathcal{P}_{3}T_{\hat{3},4}T_{\hat{1},7}.$$

The two sides act on $\mathcal{U}_{\widehat{78}} \times \mathcal{U}_{\widehat{58}} \times \mathcal{U}_{\widehat{56}} \times \mathcal{U}_{\widehat{38}} \times \mathcal{U}_{\widehat{36}} \times \mathcal{U}_{\widehat{34}} \times \mathcal{U}_{\widehat{18}} \times \mathcal{U}_{\widehat{16}} \times \mathcal{U}_{\widehat{14}} \times \mathcal{U}_{\widehat{12}}$ and map to $\mathcal{U}_{\widehat{23}} \times \mathcal{U}_{\widehat{25}} \times \mathcal{U}_{\widehat{27}} \times \mathcal{U}_{\widehat{45}} \times \mathcal{U}_{\widehat{47}} \times \mathcal{U}_{\widehat{67}}$, which can be read off from [13, Figure 19].

⁸It should be clear from the context when $\{,,\}$ denotes a ternary operation and not a set.

If all basic sets are equal and we are dealing with a single map T, this reduces to

$$T_1\mathcal{P}_4\mathcal{P}_5\mathcal{P}_6T_3\mathcal{P}_6\mathcal{P}_5\mathcal{P}_2T_3\mathcal{P}_6T_1\mathcal{P}_4\mathcal{P}_5\mathcal{P}_6\mathcal{P}_3=\mathcal{P}_3T_4\mathcal{P}_3\mathcal{P}_2\mathcal{P}_1T_3\mathcal{P}_6\mathcal{P}_2\mathcal{P}_3T_4T_7.$$

Writing

$$T(a, b, c, d) = (\{a, b, c, d\}, \langle a, b, c, d\rangle, [a, b, c, d]),$$

with three quaternary operations $\mathcal{U}^4 \to \mathcal{U}$, the octagon equation, acting on (a, b, c, d, e, f, g, h, k, l) results in the following six conditions,

$$\{\{a, b, d, g\}, \{[a, b, d, g], c, e, h\}, \{\langle a, b, d, g \rangle, \langle [a, b, d, g], c, e, h \rangle, f, k\}, l\}$$

$$= \{d, e, f, \{g, h, k, l\}\},$$

$$\langle \{a, b, d, g\}, \{[a, b, d, g], c, e, h\}, \{\langle a, b, d, g \rangle, \langle [a, b, d, g], c, e, h \rangle, f, k\}, l\rangle$$

$$= \{b, c, \langle d, e, f, \{g, h, k, l\} \rangle, \langle g, h, k, l \rangle\},$$

$$[\{a, b, d, g\}, \{[a, b, d, g], c, e, h\}, \{\langle a, b, d, g \rangle, \langle [a, b, d, g], c, e, h \rangle, f, k\}, l]$$

$$= \{a, [b, c, \langle d, e, f, \{g, h, k, l\} \rangle, \langle g, h, k, l \rangle], [d, e, f, \{g, h, k, l\}], [g, h, k, l]\},$$

$$\langle \langle a, b, d, g \rangle, \langle [a, b, d, g], c, e, h \rangle, f, k \rangle = \langle b, c, \langle d, e, f, \{g, h, k, l\} \rangle, [g, h, k, l] \rangle,$$

$$[\langle a, b, d, g \rangle, \langle [a, b, d, g], c, e, h \rangle, f, k]$$

$$= \langle a, [b, c, \langle d, e, f, \{g, h, k, l\} \rangle, \langle g, h, k, l \rangle], [d, e, f, \{g, h, k, l\}], [g, h, k, l] \rangle,$$

$$[a, [b, c, \langle d, e, f, \{g, h, k, l\} \rangle, \langle g, h, k, l \rangle], [d, e, f, \{g, h, k, l\}], [g, h, k, l] \rangle,$$

$$= [[a, b, d, g], c, e, h],$$

$$(5.13)$$

for all $a, b, c, d, e, f, g, h, k, l \in \mathcal{U}$.

6 Examples of dual polygon equations

In this section, we elaborate dual polygon equations up to the dual 8-gon equation.

6.1 Dual trigon equation

This is the equation

 $T_{12}T_{23} = T_{13}$

for maps $\tilde{T}_{ij} : \mathcal{U}_j \to \mathcal{U}_i, i < j$.

6.2 Dual tetragon equation

For i, j = 1, 2, 3, 4, i < j, let $L_{ij}: \mathcal{U}_i \to \mathcal{U}_j$ carry a parameter from a set \mathcal{U}_{ij} . Let each of the local trigon equations

$$L_{ik}(u_{ik}) = L_{jk}(v_{jk})L_{ij}(v_{ij}), \qquad i < j < k,$$

uniquely determine a map

$$\widetilde{T}_{ijk}: \ \mathcal{U}_{ik} \longrightarrow \mathcal{U}_{ij} \times \mathcal{U}_{jk}, \qquad u_{ik} \mapsto (v_{ij}, v_{jk}).$$

Then the maps \tilde{T}_{ijk} have to satisfy a consistency condition, which is obtained by reversing the arrows in Figure 1. This means that the maps \tilde{T}_{ijk} have to satisfy the dual tetragon equation,

$$\tilde{T}_{123,2}\tilde{T}_{134} = \tilde{T}_{234,1}\tilde{T}_{124},\tag{6.1}$$

which acts on \mathcal{U}_{14} .

If all basic sets are equal and there is only a single map \tilde{T} , (6.1) reduces to

$$\tilde{T}_2 \tilde{T} = \tilde{T}_1 \tilde{T}. \tag{6.2}$$

Writing $\tilde{T} = (\tilde{T}_1, \tilde{T}_2)$, this amounts to idempotency and commutativity of the maps $\tilde{T}_i : \mathcal{U} \to \mathcal{U}$, i = 1, 2.

Remark 6.1. In a framework of vector spaces, with the Cartesian product replaced by the corresponding tensor product, and linear maps, (6.2) means that \tilde{T} is *coassociative*.

6.3 Dual pentagon equation

Using complementary index notation, the dual pentagon equation is

$$\tilde{T}_{\hat{5}}\tilde{T}_{\hat{3}}\tilde{T}_{\hat{1}}=\tilde{T}_{\hat{2}}\tilde{T}_{\hat{4}},$$

for maps

$$\tilde{T}_{ijkl}: \ \mathcal{U}_{ijl} \times \mathcal{U}_{jkl} \to \mathcal{U}_{ikl} \times \mathcal{U}_{ijk}, \qquad 1 \le i < j < k < l \le 5.$$

Letting it act on $\mathcal{U}_{125} \times \mathcal{U}_{235} \times \mathcal{U}_{345}$, this equation takes the form

 $\tilde{T}_{\hat{5},\mathbf{2}}\tilde{T}_{\hat{3},\mathbf{1}}\tilde{T}_{\hat{1},\mathbf{2}} = \tilde{T}_{\hat{2},\mathbf{1}}\mathcal{P}_{\mathbf{2}}\tilde{T}_{\hat{4},\mathbf{1}}.$

If all the basic sets are the same and there is only a single map \tilde{T} , this is simply

$$\tilde{T}_2 \tilde{T}_1 \tilde{T}_2 = \tilde{T}_1 \mathcal{P}_2 \tilde{T}_1.$$

Writing $\tilde{T}(a, b) =: (a \cdot b, a * b)$, the last equation is equivalent to

$$a \cdot (b \cdot c) = (a \cdot b) \cdot c, \qquad (a * (b \cdot c)) \cdot (b * c) = (a \cdot b) * c, (a * (b \cdot c)) * (b * c) = a * b,$$
(6.3)

for all $a, b, c \in \mathcal{U}$.

6.4 Dual hexagon equation

This is the equation

$$\tilde{T}_{\hat{6}}\tilde{T}_{\hat{4}}\tilde{T}_{\hat{2}}=\tilde{T}_{\hat{1}}\tilde{T}_{\hat{3}}\tilde{T}_{\hat{5}}$$

for maps

$$\tilde{T}_{ijklm}: \ \mathcal{U}_{ijkm} \times \mathcal{U}_{iklm} \to \mathcal{U}_{jklm} \times \mathcal{U}_{ijlm} \times \mathcal{U}_{ijkl}, \qquad 1 \le i < j < k < l < m \le 6.$$

Introducing position indices and transposition maps, it takes the form

$$\tilde{T}_{\hat{6},\mathbf{1}}\tilde{T}_{\hat{4},\mathbf{2}}\mathcal{P}_{\mathbf{3}}\tilde{T}_{\hat{2},\mathbf{1}} = \mathcal{P}_{\mathbf{3}}\tilde{T}_{\hat{1},\mathbf{4}}\tilde{T}_{\hat{3},\mathbf{2}}\mathcal{P}_{\mathbf{1}}\tilde{T}_{\hat{5},\mathbf{2}}$$

If all the basic sets are the same and there is only a single map \tilde{T} , this is

$$\tilde{T}_1 \tilde{T}_2 \mathcal{P}_3 \tilde{T}_1 = \mathcal{P}_3 \tilde{T}_4 \tilde{T}_2 \mathcal{P}_1 \tilde{T}_2, \tag{6.4}$$

which already appeared as a 4-cocycle condition in [58]. Writing

 $\tilde{T}(a,b) =: (a * b, a \cdot b, a \diamond b),$

(6.4) imposes (5.4) on the first two binary operations, and requires in addition

$$(a \diamond (b \cdot c)) \ast (b \diamond c) = (a \ast b) \diamond ((a \cdot b) \ast c), \qquad (a \diamond (b \cdot c)) \cdot (b \diamond c) = (a \cdot b) \diamond c,$$

$$(a \diamond (b \cdot c)) \diamond (b \diamond c) = a \diamond b,$$

$$(6.5)$$

for all $a, b, c \in \mathcal{U}$.

Remark 6.2. In [29] (see (3.8) therein), the dual hexagon equation appeared, as a realization of a Pachner move of type (3,3) in four dimensions, in the form

$$(Q\mathcal{P})_1 Q_2 \mathcal{P}_1 Q_2 = \mathcal{P}_3 (Q\mathcal{P})_4 Q_2 \mathcal{P}_3 Q_1.$$

Indeed, setting $Q = \tilde{T}\mathcal{P}$, this reads

$$\tilde{T}_1\tilde{T}_2\mathcal{P}_2\mathcal{P}_1\tilde{T}_2\mathcal{P}_2 = \mathcal{P}_3\tilde{T}_4\tilde{T}_2\mathcal{P}_2\mathcal{P}_3\tilde{T}_1\mathcal{P}_1.$$

Writing it as

$$\tilde{T}_1\tilde{T}_2\mathcal{P}_3(\mathcal{P}_3\mathcal{P}_2\mathcal{P}_1)\tilde{T}_2\mathcal{P}_2=\mathcal{P}_3\tilde{T}_4\tilde{T}_2\mathcal{P}_1(\mathcal{P}_1\mathcal{P}_2\mathcal{P}_3)\tilde{T}_1\mathcal{P}_1$$

and using the identities $\mathcal{P}_3\mathcal{P}_2\mathcal{P}_1\tilde{T}_2 = \tilde{T}_1\mathcal{P}_2\mathcal{P}_1, \ \mathcal{P}_1\mathcal{P}_2\mathcal{P}_3\tilde{T}_1 = \tilde{T}_2\mathcal{P}_1\mathcal{P}_2$, we obtain

$$\tilde{T}_1\tilde{T}_2\mathcal{P}_3\tilde{T}_1\mathcal{P}_2\mathcal{P}_1\mathcal{P}_2 = \mathcal{P}_3\tilde{T}_4\tilde{T}_2\mathcal{P}_1\tilde{T}_2\mathcal{P}_1\mathcal{P}_2\mathcal{P}_1,$$

which, by use of the braid equation $\mathcal{P}_2\mathcal{P}_1\mathcal{P}_2 = \mathcal{P}_1\mathcal{P}_2\mathcal{P}_1$, is equivalent to (6.4).

6.5 Dual heptagon equation

The dual heptagon equation is

$$\mathcal{P}_{3}\tilde{T}_{\hat{7},4}\tilde{T}_{\hat{5},2}\mathcal{P}_{4}\mathcal{P}_{1}\tilde{T}_{\hat{3},2}\tilde{T}_{\hat{1},4} = \tilde{T}_{\hat{2},1}\mathcal{P}_{3}\mathcal{P}_{4}\tilde{T}_{\hat{4},2}\mathcal{P}_{5}\mathcal{P}_{4}\mathcal{P}_{3}\tilde{T}_{\hat{6},1}\mathcal{P}_{3}, \tag{6.6}$$

with maps \tilde{T}_{ijklmp} : $\mathcal{U}_{ijklp} \times \mathcal{U}_{ijlmp} \times \mathcal{U}_{jklmp} \rightarrow \mathcal{U}_{iklmp} \times \mathcal{U}_{ijkmp} \times \mathcal{U}_{ijklm}$, i < j < k < l < m < p. Both sides act on $\mathcal{U}_{12347} \times \mathcal{U}_{12457} \times \mathcal{U}_{12567} \times \mathcal{U}_{23457} \times \mathcal{U}_{23567} \times \mathcal{U}_{34567}$ and map to $\mathcal{U}_{14567} \times \mathcal{U}_{13467} \times \mathcal{U}_{13456} \times \mathcal{U}_{12367} \times \mathcal{U}_{12345}$.

If all basic sets are the same and there is only a single map \tilde{T} , writing

$$T(a,b,c) = (\{a,b,c\}, \langle a,b,c\rangle, [a,b,c]),$$

(6.6) is equivalent to the following conditions for the three ternary operations,

$$\{b, c, \{d, e, f\}\} = \{\{a, b, d\}, \{\langle a, b, d \rangle, c, e\}, f\}, \\ \{a, \langle b, c, \{d, e, f\} \rangle, \langle d, e, f \rangle\} = \langle \{a, b, d\}, \{\langle a, b, d \rangle, c, e\}, f \rangle, \\ \langle a, \langle b, c, \{d, e, f\} \rangle, \langle d, e, f \rangle \rangle = \langle \langle a, b, d \rangle, c, e \rangle, \\ \{[a, \langle b, c, \{d, e, f\} \rangle, \langle d, e, f \rangle], [b, c, \{d, e, f\}], [d, e, f]\} = [\{a, b, d\}, \{\langle a, b, d \rangle, c, e\}, f], \\ \langle [a, \langle b, c, \{d, e, f\} \rangle, \langle d, e, f \rangle], [b, c, \{d, e, f\}], [d, e, f] \rangle = [\langle a, b, d \rangle, c, e], \\ [[a, \langle b, c, \{d, e, f\} \rangle, \langle d, e, f \rangle], [b, c, \{d, e, f\}], [d, e, f]] = [a, b, d],$$

$$(6.7)$$

for all $a, b, c, d, e, f \in \mathcal{U}$.

6.6 Dual octagon equation

The dual octagon equation is

$$\mathcal{P}_{4}\mathcal{P}_{7}\mathcal{P}_{5}\mathcal{P}_{6}\tilde{T}_{\hat{8},7}\mathcal{P}_{3}\tilde{T}_{\hat{6},4}\mathcal{P}_{6}\mathcal{P}_{3}\mathcal{P}_{2}\tilde{T}_{\hat{4},3}\mathcal{P}_{1}\mathcal{P}_{2}\mathcal{P}_{3}\tilde{T}_{\hat{2},4} = \tilde{T}_{\hat{1},1}\tilde{T}_{\hat{3},3}\mathcal{P}_{5}\mathcal{P}_{6}\mathcal{P}_{2}\tilde{T}_{\hat{5},3}\mathcal{P}_{6}\mathcal{P}_{5}\mathcal{P}_{4}\tilde{T}_{\hat{7},1}\mathcal{P}_{3},$$

for maps $\tilde{T}_{ijklmpq} : \mathcal{U}_{ijklmpq} \times \mathcal{U}_{ijkmpq} \times \mathcal{U}_{iklmpq} \to \mathcal{U}_{jklmpq} \times \mathcal{U}_{ijlmpq} \times \mathcal{U}_{ijklpq} \times \mathcal{U}_{ijklmp}$, where i < j < k < l < m < p < q. Both sides of the equation act on $\mathcal{U}_{\widehat{67}} \times \mathcal{U}_{\widehat{47}} \times \mathcal{U}_{\widehat{45}} \times \mathcal{U}_{\widehat{27}} \times \mathcal{U}_{\widehat{25}} \times \mathcal{U}_{\widehat{23}}$.

If all basic sets are the same and there is only a single map T, writing

$$T(a, b, c) = (\{a, b, c\}, \langle a, b, c \rangle, [a, b, c], |a, b, c|)$$

with four ternary operations $\mathcal{U}^3 \to \mathcal{U}$, the dual octagon equation, acting on (a, b, c, d, e, f) results in (5.10), (5.12), and the following conditions,

$$\{ |a, [b, c, \langle d, e, f \rangle], [d, e, f]|, |b, c, \langle d, e, f \rangle|, |d, e, f| \}$$

$$= |\{a, b, d\}, \{ [a, b, d], c, e\}, \{ \langle a, b, d \rangle, \langle [a, b, d], c, e\} \rangle, f|,$$

$$\langle |a, [b, c, \langle d, e, f \rangle], [d, e, f]|, |b, c, \langle d, e, f \rangle|, |d, e, f| \rangle = |\langle a, b, d \rangle, \langle [a, b.d], c, e, \rangle, f|,$$

$$[|a, [b, c, \langle d, e, f \rangle], [d, e, f]|, |b, c, \langle d, e, f \rangle|, |d, e, f|] = |[a, b, d], c, e|,$$

$$||a, [b, c, \langle d, e, f \rangle], [d, e, f]|, |b, c, \langle d, e, f \rangle|, |d, e, f|] = |a, b, d|.$$

$$(6.8)$$

7 Relations between solutions of neighboring polygon equations

By a (dual) polygon map we mean a solution of a (dual) polygon equation. In this section, the case when all the basic sets appearing in a multiple Cartesian product, on which a (dual) polygon map acts, are the same set \mathcal{U} is considered.

First, we formulate a few special cases of theorems in Section 4. Though they are rather corollaries, because of their relevance they also deserve to be called theorems. Let $N \in \mathbb{N}$, N > 2.

Theorem 7.1. Let $\tilde{T}^{(N+1)}$ be a dual (N+1)-gon map and T that map with the last component of its codomain cut off. Then T is an N-gon map.

Theorem 7.2. For even N, let $T^{(N+1)}$ be an (N+1)-gon map and T that map with the first component of its codomain cut off. Then T is an N-gon map.

Theorem 7.3. For odd N, let $\tilde{T}^{(N+1)}$ be a dual (N+1)-gon map and \tilde{T} that map with the first component of its codomain cut off. Then \tilde{T} is a dual N-gon map.

Further theorems derived from those in Section 4 will be formulated in the following subsections, where we also provide examples for these results and derive more powerful results for the (dual) polygon equations up to the (dual) octagon equation.

7.1 Degenerate dual (N + 1)-gon maps from (dual) N-gon maps

For any (not necessarily trigon or dual trigon) map $T: \mathcal{U} \to \mathcal{U}, T^{(4)}(a, b) := T(a)$ and also $T^{(4)}(a, b) := T(b)$ are (degenerate) tetragon maps. Furthermore, we have the following.

Proposition 7.4. Let $\tilde{T}^{(4)}$ be a dual tetragon map. Then $T^{(5)}(a,b) := \tilde{T}^{(4)}(b)$ is a pentagon map and $\tilde{T}^{(5)}(a,b) := \tilde{T}^{(4)}(a)$ is a dual pentagon map.

Proof. This is quickly verified.

Proposition 7.5. Let $T^{(5)}$ be a pentagon map. Then $T^{(6)}(a, b, c) := T^{(5)}(a, b)$ is a hexagon map.

Proof. If $\langle a, b, c \rangle = a * b$ and $[a, b, c] = a \cdot b$, (5.10) becomes (5.4).

Proposition 7.6. Let $\tilde{T}^{(5)}$ be a dual pentagon map. Then $T^{(6)}(a, b, c) := \tilde{T}^{(5)}(b, c)$ is a hexagon map.

Proof. If $\langle a, b, c \rangle = b \cdot c$ and [a, b, c] = b * c, then (5.10) becomes (6.3).

Proposition 7.7. Let $\tilde{T}^{(6)}$ be a dual hexagon map. Then $T^{(7)}(a, b, c) := \tilde{T}^{(6)}(b, c)$ is a heptagon map.

Proof. Writing $T^{(7)}(a, b, c) = (b * c, b \circ c)$, the first two of equations (5.12) become the first two of (5.4). The last two of (5.10) become the last two of (6.5). The first of equations (5.10) becomes the third of (5.4), the third of (5.12) the first of (6.5).

Proposition 7.8. Let $\tilde{T}^{(6)}$ be a dual hexagon map. Then $\tilde{T}^{(7)}(a, b, c) := \tilde{T}^{(6)}(a, b)$ is a dual heptagon map.

Proof. Writing $\tilde{T}^{(7)}(a, b, c) = (a * b, a \cdot b, a \diamond b), (6.7)$ becomes (5.4) and (6.5).

Proposition 7.9. Let $T^{(7)}$ be a heptagon map. Then $T^{(8)}(a, b, c, d) := T^{(7)}(a, b, c)$ is an octagon map.

Proof. If T does not depend on the last argument, setting

$$\{a, b, c, d\} =: \{a, b, c\}, \qquad \langle a, b, c, d\rangle =: \langle a, b, c\rangle, \qquad [a, b, c, d] =: [a, b, c],$$

the first three of conditions (5.13) reduce to (5.12) and the last three to (5.10). The resulting six conditions are those for $T^{(7)}$ to be a heptagon map.

These results suggest that, more generally, the following statements hold, which determine extensions of each solution of an odd N-gon equation, or each solution of a dual N-gon equation, to a degenerate solution of a (dual) (N + 1)-gon equation. Indeed, these general results follow from theorems in Section 4.

Theorem 7.10. For $n \in \mathbb{N}$, n > 1, let $T^{(2n-1)}$ be a (2n-1)-gon map. Then

$$T^{(2n)}(a_1,\ldots,a_n) := T^{(2n-1)}(a_1,\ldots,a_{n-1})$$

is a 2n-gon map.

Proof. This a special case of Theorem 4.1(2). See the proof there.

Theorem 7.11. For $n \in \mathbb{N}$, n > 1, let $\tilde{T}^{(2n-1)}$ be a dual (2n-1)-gon map. Then

$$T^{(2n)}(a_1,\ldots,a_n) := \tilde{T}^{(2n-1)}(a_2,\ldots,a_n)$$

is a 2n-gon map.

Proof. This a special case of Theorem 4.7(2).

Theorem 7.12. For $n \in \mathbb{N}$, n > 1, let $\tilde{T}^{(2n)}$ be a dual 2n-gon map. Then

$$T^{(2n+1)}(a_1,\ldots,a_n) := \tilde{T}^{(2n)}(a_2,\ldots,a_n)$$

is a (2n+1)-gon map.

Proof. This a special case of Theorem 4.7(2).

Theorem 7.13. For $n \in \mathbb{N}$, n > 1, let $\tilde{T}^{(2n)}$ be a dual 2n-gon map. Then

 $\tilde{T}^{(2n+1)}(a_1,\ldots,a_n) := \tilde{T}^{(2n)}(a_1,\ldots,a_{n-1})$

is a dual (2n+1)-gon map.

Proof. This a special case of Theorem 4.5(2).

7.2 Further relations between neighboring polygon maps

Proposition 7.14. Let $T_i^{(3)}$, i = 1, 2, be trigon maps. Then

$$\tilde{T}^{(4)}(a) = \left(T_1^{(3)}(a), T_2^{(3)}(a)\right)$$

is a dual tetragon map if and only if the two trigon maps commute.

Proof. This is easily verified. Also see Section 6.2.

Proposition 7.15. For a map $T^{(5)}: \mathcal{U} \times \mathcal{U} \to \mathcal{U} \times \mathcal{U}$, let us write

 $T^{(5)}(a,b) =: (a * b, T^{(4)}(a,b)),$

with a binary operation * and a map $T^{(4)}: \mathcal{U} \times \mathcal{U} \to \mathcal{U}$. The following conditions are equivalent:

- (1) $T^{(5)}$ is a pentagon map.
- (2) $T^{(4)}$ is a tetragon map and, with $a \cdot b := T^{(4)}(a, b)$, the binary operations \cdot and * satisfy

$$(a * b) * ((a \cdot b) * c) = b * c,$$
 $(a * b) \cdot ((a \cdot b) * c) = a * (b \cdot c),$

for all $a, b, c \in \mathcal{U}$.

Proof. (5.4) shows that the conditions for $T^{(5)}$ to be a pentagon map are equivalent to $T^{(4)}$ being a tetragon map (which means associativity of \cdot) and the two compatibility conditions for the two binary operations.

Corollary 7.16. Let $T^{(4)}$ be a tetragon map. Then

$$T^{(5)}(a,b) = (b, T^{(4)}(a,b))$$

is a pentagon map.

Proof. Setting a * b = b solves the two compatibility equations in condition (2) of the preceding proposition.

Corollary 7.17. Let $T^{(4)}$ be a tetragon map and u a fixed element of \mathcal{U} . Then

$$T^{(5)}(a,b) = (u, T^{(4)}(a,b))$$

is a pentagon map if and only if $T^{(4)}(u, u) = u$.

Proof. Setting a * b = u for all $a, b \in \mathcal{U}$, solves the first of the two compatibility equations in condition (2) of the preceding proposition and reduces the second to $u \cdot u = u$.

Proposition 7.18. Let $T^{(4)}$ be a tetragon map.

(1) The map

$$\tilde{T}^{(5)}(a,b) = (T^{(4)}(a,b),a)$$

is a dual pentagon map.

(2) If u is a fixed element of \mathcal{U} , then

$$\tilde{T}^{(5)}(a,b) = (T^{(4)}(a,b),u)$$

is a dual pentagon map if and only if $T^{(4)}(u, u) = u$.

Proof. This is easily verified.

Proposition 7.19. For a map $\tilde{T}^{(6)}: \mathcal{U} \times \mathcal{U} \to \mathcal{U} \times \mathcal{U}$, let us write

$$\tilde{T}^{(6)}(a,b) \coloneqq (T^{(5)}(a,b), a \diamond b),$$

with a map $T^{(5)}: \mathcal{U} \times \mathcal{U} \to \mathcal{U} \times \mathcal{U}$ and a binary operation \diamond . The following conditions are equivalent:

- (1) $\tilde{T}^{(6)}$ is a dual hexagon map.
- (2) $T^{(5)}$ is a pentagon map and, expressed as in (5.3) (so that (5.4) holds), it satisfies (6.5).

Proof. This is easily verified.

Example 7.20. Let \cdot be commutative and $a \diamond b := b * a$. Then, as a consequence of (5.4), the additional dual hexagon map conditions (6.5) are reduced to the single condition

$$((a \cdot b) * c) * (a * b) = ((b \cdot c) * a) * (c * b).$$

This holds, for example, if $\mathcal{U} = (0,1) \subset \mathbb{R}$ and a * b := (1-a)b/(1-ab). Hence, using the pentagon map (5.7), the map

$$(a,b) \longmapsto \left(\frac{(1-a)b}{1-ab}, ab, \frac{a(1-b)}{1-ab}\right)$$

solves the dual hexagon equation, also see [29]. This map actually shows up in an identity for the *Rogers dilogarithm* function L(a) [52]. $S(a) = e^{\lambda L(a)}$, with an arbitrary constant $\lambda \neq 0$, satisfies the pentagon relation⁹

$$S(b)S(a) = S\left(\frac{a(1-b)}{1-ab}\right)S(ab)S\left(\frac{(1-a)b}{1-ab}\right)$$

Kashaev called a solution $\hat{T}: I \to \text{End}(\mathcal{U} \otimes \mathcal{U})$, where I is the open unit interval $(0, 1) \subset \mathbb{R}$ and \mathcal{U} a vector space, a matrix or operator dilogarithm if it satisfies the local pentagon equation [26, 27]

$$\hat{T}_{23}(a)\hat{T}_{12}(b) = \hat{T}_{12}\left(\frac{a(1-b)}{1-ab}\right)\hat{T}_{13}(ab)\hat{T}_{23}\left(\frac{(1-a)b}{1-ab}\right)$$

Corollary 7.21. If $T^{(5)}$ is a pentagon map, then

$$\tilde{T}^{(6)}(a,b) := (T^{(5)}(a,b),a)$$

is a dual hexagon map.

Proof. Setting $a \diamond b := a$ solves the three equations (6.5).

Corollary 7.22. Let $T^{(5)}$ be a pentagon map and $u \in \mathcal{U}$ a fixed element. Then

$$\tilde{T}^{(6)}(a,b) := (T^{(5)}(a,b),u)$$

is a dual hexagon map if and only if $T^{(5)}(u, u) = (u, u)$.

Proof. Setting $a \diamond b := u$ reduces the three equations (6.5) to u * u = u and $u \cdot u = u$.

⁹If we order the parameters as $(a_0, \ldots, a_4) := (a, 1-ab, b, (1-b)/(1-ab), (1-a)/(1-ab))$, they are given by the recursion relation $a_{n-1}a_{n+1} = 1 - a_n$ (a special Y-system), which has \mathbb{Z}_5 symmetry $a_{n+5} = a_n$ [18, 20, 66].

Proposition 7.23. Let $\tilde{T}^{(5)}$ be a dual pentagon map.

(1) The map

$$\tilde{T}^{(6)}(a,b) = (b, \tilde{T}^{(5)}(a,b))$$

is a dual hexagon map.

(2) If u is a fixed element of \mathcal{U} , then

$$\tilde{T}^{(6)}(a,b) = (u, \tilde{T}^{(5)}(a,b))$$

is a dual hexagon map if and only if $\tilde{T}^{(5)}(u, u) = (u, u)$.

Proof. This is also easily verified using results of Section 6.

Proposition 7.24. For a map $T^{(7)}: \mathcal{U} \times \mathcal{U} \times \mathcal{U} \to \mathcal{U} \times \mathcal{U} \times \mathcal{U}$, let us write

$$T^{(7)}(a,b,c) =: (\{a,b,c\}, T^{(6)}(a,b,c)),$$

with a ternary operation $\{,,\}$ and a map $T^{(6)}: \mathcal{U} \times \mathcal{U} \times \mathcal{U} \to \mathcal{U} \times \mathcal{U}$. The following conditions are equivalent:

- (1) $T^{(7)}$ is a heptagon map.
- (2) $T^{(6)}$ is a hexagon map and, expressed as in (5.9) (so that (5.10) holds), it satisfies the compatibility conditions (5.12) with the above ternary operation.

Proof. This is an immediate consequence of the last part of Section 5.5.

Corollary 7.25. If $T^{(6)}$ is a hexagon map, then

$$T^{(7)}(a,b,c) := (c,T^{(6)}(a,b,c))$$

is a heptagon map.

Proof. Using $\{a, b, c\} = c$ in (5.12) results in identities.

Corollary 7.26. Let $T^{(6)}$ be a hexagon map and $u \in \mathcal{U}$ a fixed element. Then

$$T^{(7)}(a,b,c) := \left(u, T^{(6)}(a,b,c)\right)$$

is a heptagon map if and only if $T^{(6)}(u, u, u) = (u, u)$.

Proof. Using $\{a, b, c\} = u$ in (5.12) results in $\langle u, u, u \rangle = u = [u, u, u]$.

Proposition 7.27. Let $T^{(6)}$ be a hexagon map.

(1) The map

$$\tilde{T}^{(7)}(a,b,c) = \left(T^{(6)}(a,b,c),a\right)$$

is a dual heptagon map.

(2) If u is a fixed element of \mathcal{U} , then

$$\tilde{T}^{(7)}(a,b,c) = \left(T^{(6)}(a,b,c), u\right)$$

is a dual heptagon map if and only if $T^{(6)}(u, u, u) = (u, u)$.

Proof. (1) Setting [a, b, c] = a, the last three equations of (6.7) become identities and the first three are equivalent to (5.10) by a renaming of the ternary operations.

(2) With [a, b, c] = u, the last three of equations (6.7) become $T^{(6)}(u, u, u) = (u, u)$.

Proposition 7.28. For a map $\tilde{T}^{(8)}: \mathcal{U} \times \mathcal{U} \times \mathcal{U} \to \mathcal{U} \times \mathcal{U} \times \mathcal{U}$, let us write

 $\tilde{T}^{(8)}(a,b,c) =: (T^{(7)}(a,b,c), |a,b,c|),$

with a map $T^{(7)}: \mathcal{U} \times \mathcal{U} \times \mathcal{U} \to \mathcal{U} \times \mathcal{U} \times \mathcal{U}$ and a ternary operation |, |. The following conditions are equivalent:

- (1) $\tilde{T}^{(8)}$ is a dual hexagon map.
- (2) $T^{(7)}$ is a heptagon map and, expressed as in (5.11), it satisfies (6.8).

Proof. This immediately follows from results in Section 6.6.

Corollary 7.29. If $T^{(7)}$ is a heptagon map, then

$$\tilde{T}^{(8)}(a,b,c) := (T^{(7)}(a,b,c),a)$$

is a dual octagon map.

Proof. Setting |a, b, c| = a turns the four equations (6.8) into identities.

Corollary 7.30. Let $T^{(7)}$ be a heptagon map and $u \in \mathcal{U}$ a fixed element. Then

$$\tilde{T}^{(8)}(a,b,c) := (T^{(7)}(a,b,c),u)$$

is a dual octagon map if and only if $T^{(7)}(u, u, u) = (u, u, u)$.

Proof. Setting |a, b, c| = u turns the four equations (6.8) into $\{u, u, u\} = u$, $\langle u, u, u \rangle = u$ and [u, u, u] = u.

Proposition 7.31. Let $\tilde{T}^{(7)}$ be a dual heptagon map.

(1) The map

$$\tilde{T}^{(8)}(a,b,c) = (c,\tilde{T}^{(7)}(a,b,c))$$

is a dual octagon map.

(2) If u is a fixed element of \mathcal{U} , then

$$\tilde{T}^{(8)}(a,b,c) = \left(u, \tilde{T}^{(7)}(a,b,c)\right)$$

is a dual octagon map if and only if $\tilde{T}^{(7)}(u, u, u) = (u, u, u)$.

Proof. This can be verified using results of Section 6.

Preceding results suggest the following conjectures.

Conjecture 7.32. Let $T^{(2n)}$ be a 2n-gon map, $n \in \mathbb{N}$, $n \geq 2$. Then

$$T^{(2n+1)}(a_1,\ldots,a_n) := (a_n, T^{(2n)}(a_1,\ldots,a_n))$$

is a (2n+1)-gon map.

Conjecture 7.33. Let $T^{(2n)}$ be a 2n-gon map, $n \in \mathbb{N}$, and $u \in \mathcal{U}$ a fixed element. Then

$$T^{(2n+1)}(a_1,\ldots,a_n) := (u,T^{(2n)}(a_1,\ldots,a_n))$$

is a (2n+1)-gon map if and only if $T^{(2n)}(u, ..., u) = (u, ..., u)$.

Conjecture 7.34. Let $T^{(2n+1)}$ be a (2n+1)-gon map, $n \in \mathbb{N}$. Then

$$\tilde{T}^{(2n+2)}(a_1,\ldots,a_n) := (T^{(2n+1)}(a_1,\ldots,a_n),a_1)$$

is a dual (2n+2)-gon map.

Conjecture 7.35. Let $T^{(2n+1)}$ be a (2n+1)-gon map, $n \in \mathbb{N}$, and $u \in \mathcal{U}$ a fixed element. Then

$$\tilde{T}^{(2n+2)}(a_1,\ldots,a_n) := (T^{(2n+1)}(a_1,\ldots,a_n),u)$$

is a dual (2n+2)-gon map if and only if $T^{(2n+1)}(u, ..., u) = (u, ..., u)$.

Conjecture 7.36. Let $\tilde{T}^{(2n+1)}$ be a dual (2n+1)-gon map, $n \in \mathbb{N}$. Then

$$\tilde{T}^{(2n+2)}(a_1,\ldots,a_n) := (a_n,\tilde{T}^{(2n+1)}(a_1,\ldots,a_n))$$

is a dual (2n+2)-gon map.

Conjecture 7.37. Let $\tilde{T}^{(2n+1)}$ be a dual (2n+1)-gon map, $n \in \mathbb{N}$, and $u \in \mathcal{U}$ a fixed element. Then

 $\tilde{T}^{(2n+2)}(a_1,\ldots,a_n) := (u,\tilde{T}^{(2n+1)}(a_1,\ldots,a_n))$

is a dual (2n+2)-gon map if and only if $\tilde{T}^{(2n+1)}(u, \ldots, u) = (u, \ldots, u)$.

8 Conclusions

The main results of this work concern the structure of solutions of polygon equations. More precisely, we have shown that a solution of a (dual) N-gon equation is related in very simple ways to solutions of the (dual) (N + 1)-gon and (dual) (N - 1)-gon equation. Each solution of a (dual) polygon equation extends to solutions of the higher equations.

For a chosen polygon equation, the most important case is when all basic sets are equal and there is only a single polygon map. Expressing polygon equations with the help of transposition maps, we can quite easily verify the above mentioned features for the simplest equations of the family. But for general proofs, we had to return to the underlying framework of higher Tamari orders, as developed in [13].

Our results reveal a beautiful structure of the family of polygon equations. Other nice aspects are the integrability feature [13], recalled in some examples in Section 5, and relations with polyhedra [13].

In this work, we concentrated on the set-theoretic setting. Many results directly pass over to the framework of vector spaces, tensor products, and linear maps, where the dual tetragon equation becomes the coassociativity condition and the pentagon equation plays one of its most important roles, as mentioned in the introduction. A further exploration of polygon equations, beyond the pentagon equation, in this framework, will be left for a separate work.

Also, concerning set-theoretic solutions, we have only set the stage. It is to be expected that more concrete solutions can be obtained by applying methods that have already been exploited in the case of the pentagon equation. Whereas the (dual) pentagon and dual hexagon equation can be expressed in terms of binary operations, the hexagon and higher polygon equations involve *n*-ary operations with n > 2. There is a vast literature about such generalizations of products, and this should be helpful in finding solutions of *N*-gon equations with N > 5. In particular, there are corresponding generalizations of co-, bi- and Hopf algebras, for which higher polygon equations may play a role. Also see [58] ("cocycloids").

The question pops up whether there are higher order counterparts of the important relation between solutions of the pentagon equation and bi- or Hopf algebras, mentioned in the introduction. Since the transposition \mathcal{P} solves the pentagon equation and also its dual, Corollaries 7.22 and 7.23 show that $\mathcal{P} \otimes u$ and $u \otimes \mathcal{P}$, where u is a fixed element, are both dual hexagon maps. If \mathcal{U} is an algebra \mathcal{A} , choosing $u = 1_{\mathcal{A}}$, these are algebra homomorphisms. They can be regarded as counterparts of the trivial comultiplications Δ_{ℓ} and Δ_r mentioned in the introduction. It should be obvious how this extends to higher dual polygon equations, since a special solution of the odd N-gon equation is a combination of \mathcal{P} 's that achieves a total inversion. For example, since $\mathcal{P}_{inv} := \mathcal{P}_1 \mathcal{P}_2 \mathcal{P}_1$ solves the heptagon equation and its dual, it follows that $\mathcal{P}_{inv} \otimes 1_{\mathcal{A}}$, as well as $1_{\mathcal{A}} \otimes \mathcal{P}_{inv}$, solve the dual octagon equation. A similarity transformation, analogous to the construction of non-trivial comultiplications from a trivial one, as mentioned in the introduction, leads to a new algebra homomorphism. But under what conditions does it satisfy the respective (dual) even polygon equation?

As already mentioned in the introduction, solutions of simplex equations can be obtained from solutions of a polygon equation and its dual [13, 31, 32, 41, 53]. But an additional compatibility condition has to be solved, which needs further exploration. We plan to study this in a separate work.

We have seen in Section 7.2 that a solution of the dual hexagon equation shows up in a pentagonal relation for the (exponentiated) Rogers dilogarithm. Also higher (dual) polygon equations may play a role in this context [65]. Surely there is much more to be revealed.

Acknowledgements

Some important insights that led to this work originated from my collaboration with Aristophanes Dimakis, who, sadly, passed away in 2021. I would like to thank Jim Stasheff for helpful correspondence.

References

- Baaj S., Skandalis G., Unitaires multiplicatifs et dualité pour les produits croisés de C*-algèbres, Ann. Sci. École Norm. Sup. (4) 26 (1993), 425–488.
- [2] Baratin A., Freidel L., A 2-categorical state sum model, J. Math. Phys. 56 (2015), 011705, 18 pages, arXiv:1409.3526.
- Barrett J.W., Crane L., An algebraic interpretation of the Wheeler–DeWitt equation, *Classical Quantum Gravity* 14 (1997), 2113–2121, arXiv:gr-qc/9609030.
- [4] Biedenharn L.C., Louck J.D., Angular momentum in quantum physics. Theory and application, *Encycl. Math. Appl.*, Vol. 8, Addison-Wesley Publishing Co., Reading, MA, 1981.
- [5] Catino F., Mazzotta M., Miccoli M.M., Set-theoretical solutions of the pentagon equation on groups, Comm. Algebra 48 (2020), 83–92, arXiv:1902.04310.
- [6] Catino F., Mazzotta M., Stefanelli P., Set-theoretical solutions of the Yang–Baxter and pentagon equations on semigroups, *Semigroup Forum* 101 (2020), 259–284, arXiv:1910.05393.
- [7] Colazzo I., Jespers E., Kubat L., Set-theoretic solutions of the pentagon equation, Commun. Math. Phys. 380 (2020), 1003–1024, arXiv:2004.04028.
- [8] Crane L., Frenkel I.B., Four-dimensional topological quantum field theory, Hopf categories, and the canonical bases, J. Math. Phys. 35 (1994), 5136–5154, arXiv:hep-th/9405183,.
- [9] Davydov A.A., Pentagon equation and matrix bialgebras, Comm. Algebra 29 (2001), 2627–2650, arXiv:math.QA/0001095.

- [10] Dimakis A., Korepanov I.G., Grassmannian-parameterized solutions to direct-sum polygon and simplex equations, J. Math. Phys. 62 (2021), 051701, 17 pages, arXiv:2009.02352.
- [11] Dimakis A., Müller-Hoissen F., KP line solitons and Tamari lattices, J. Phys. A 44 (2011), 025203, 49 pages, arXiv:1009.1886.
- [12] Dimakis A., Müller-Hoissen F., KP solitons, higher Bruhat and Tamari orders, in Associahedra, Tamari Lattices and Related Structures, Progr. Math., Vol. 299, Birkhäuser, Basel, 2012, 391–423, arXiv:1110.3507.
- [13] Dimakis A., Müller-Hoissen F., Simplex and polygon equations, SIGMA 11 (2015), 042, 49 pages, arXiv:1409.7855.
- [14] Dimakis A., Müller-Hoissen F., Matrix Kadomtsev–Petviashvili equation: tropical limit, Yang–Baxter and pentagon maps, *Theoret. and Math. Phys.* **196** (2018), 1164–1173, arXiv:1709.09848.
- [15] Doliwa A., Sergeev S.M., The pentagon relation and incidence geometry, J. Math. Phys. 55 (2014), 063504, 21 pages, arXiv:1108.0944.
- [16] Edelman P.H., Reiner V., The higher Stasheff-Tamari posets, *Mathematika* 43 (1996), 127–154.
- [17] Etingof P., Schiffmann O., Lectures on quantum groups, Lect. Math. Phys., International Press, Boston, MA, 1998.
- [18] Faddeev L.D., Volkov pentagon for the modular quantum dilogarithm, Funct. Anal. Appl. 45 (2011), 291– 296, arXiv:1201.6464.
- [19] Faddeev L.D., Kashaev R.M., Quantum dilogarithm, Modern Phys. Lett. A 9 (1994), 427–434, arXiv:hep-th/9310070.
- [20] Gliozzi F., Tateo R., ADE functional dilogarithm identities and integrable models, *Phys. Lett. B* 348 (1995), 84–88, arXiv:hep-th/9411203.
- [21] Grünbaum B., Configurations of points and lines, Grad. Stud. Math., Vol. 103, American Mathematical Society, Providence, RI, 2009.
- [22] Jiang L., Liu M., On set-theoretical solution of the pentagon equation, Adv. Math. (China) **34** (2005), 331–337.
- [23] Kapranov M.M., Voevodsky V.A., Combinatorial-geometric aspects of polycategory theory: pasting schemes and higher Bruhat orders (list of results), *Cahiers Topologie Géom. Différentielle Catég.* 32 (1991), 11–27.
- [24] Kashaev R.M., Quantum dilogarithm as a 6*j*-symbol, Modern Phys. Lett. A 9 (1994), 3757–3768, arXiv:hep-th/9411147.
- [25] Kashaev R.M., The Heisenberg double and the pentagon relation, St. Petersburg Math. J. 8 (1997), 585–592, arXiv:q-alg/9503005.
- [26] Kashaev R.M., On matrix generalizations of the dilogarithm, Theoret. and Math. Phys. 118 (1999), 314–318.
- [27] Kashaev R.M., The pentagon equation and mapping class groups of surfaces with marked points, *Theoret. and Math. Phys.* **123** (2000), 576–581.
- [28] Kashaev R.M., Fully noncommutative discrete Liouville equation, in Infinite Analysis 2010—Developments in Quantum Integrable Systems, *RIMS Kôkyûroku Bessatsu*, Vol. B28, Res. Inst. Math. Sci. (RIMS), Kyoto, 2011, 89–98.
- [29] Kashaev R.M., On realizations of Pachner moves in 4d, J. Knot Theory Ramifications 24 (2015), 1541002, 13 pages, arXiv:1504.01979.
- [30] Kashaev R.M., Reshetikhin N., Symmetrically factorizable groups and self-theoretical solutions of the pentagon equation, in Quantum Groups, *Contemp. Math.*, Vol. 433, American Mathematical Society, Providence, RI, 2007, 267–279, arXiv:math.QA/0111171.
- [31] Kashaev R.M., Sergeev S.M., On pentagon, ten-term, and tetrahedron relations, Comm. Math. Phys. 195 (1998), 309–319, arXiv:q-alg/9607032.
- [32] Kassotakis P., Matrix factorizations and pentagon maps, Proc. A. 479 (2023), 20230276, 15 pages, arXiv:2302.02889.
- [33] Kelly G.M., On MacLane's conditions for coherence of natural associativities, commutativities, etc, J. Algebra 1 (1964), 397–402.
- [34] Korepanov I.G., Algebraic formulas whose structure imitates Pachner moves and new types of acyclic complexes, *Topology Atlas Invited Contributions* 7 (2002), 3 pages, arXiv:math.GT/0312458.
- [35] Korepanov I.G., Quadratic heptagon cohomology, arXiv:2110.08780.

- [36] Korepanov I.G., Heptagon relation in a direct sum, St. Petersburg Math. J. 33 (2022), 675–686, arXiv:2003.10335.
- [37] Korepanov I.G., Odd-gon relations and their cohomology, arXiv:2205.00405.
- [38] Kustermans J., Vaes S., The operator algebra approach to quantum groups, Proc. Natl. Acad. Sci. USA 97 (2000), 547–552.
- [39] Lickorish W.B.R., Simplicial moves on complexes and manifolds, in Proceedings of the Kirbyfest (Berkeley, CA, 1998), Geom. Topol. Monogr., Vol. 2, Geom. Topol. Publ., Coventry, 1999, 299–320, arXiv:math.GT/9911256.
- [40] Mac Lane S., Natural associativity and commutativity, *Rice Univ. Stud.* **49** (1963), 28–46.
- [41] Maillet J., On pentagon and tetrahedron equationsy, St. Petersburg Math. J. 6 (1995), 375–383, arXiv:hep-th/9312037.
- [42] Manin Yu.I., Schechtman V.V., Arrangements of real hyperplanes and Zamolodchikov equations, in Group Theoretical Methods in Physics, Vol. I (Yurmala, 1985), VNU Science Press, Utrecht, 1986, 151–165.
- [43] Manin Yu.I., Schechtman V.V., Higher Bruhat orders, related to the symmetric group, *Funct. Anal. Appl.* 20 (1986), 148–150.
- [44] Markl M., Models for operads, Comm. Algebra 24 (1996), 1471–1500, arXiv:hep-th/9411208.
- [45] Mazzotta M., Idempotent set-theoretical solutions of the pentagon equation, Boll. Unione Mat. Ital. 17 (2024), 457–469, arXiv:2301.01643.
- [46] Mazzotta M., Pérez-Calabuig V., Stefanelli P., Set-theoretical solutions of the pentagon equation on Clifford semigroups, *Semigroup Forum* 108 (2024), 413–431, arXiv:2301.09944.
- [47] Mihalache S., Suzuki S., Terashima Y., Quantum invariants of closed framed 3-manifolds based on ideal triangulations, arXiv:2209.07378.
- [48] Militaru G., Heisenberg double, pentagon equation, structure and classification of finite-dimensional Hopf algebras, J. London Math. Soc. (2) 69 (2004), 44–64, arXiv:math.QA/0009141.
- [49] Moore G., Seiberg N., Classical and quantum conformal field theory, Comm. Math. Phys. 123 (1989), 177–254.
- [50] Pachner U., P.L. homeomorphic manifolds are equivalent by elementary shellings, *European J. Combin.* 12 (1991), 129–145.
- [51] Ponzano G., Regge T., Semiclassical limit of Racah coefficients, in Spectroscopic and Group Theoretical Methods in Physics, North-Holland, Amsterdam, 1968, 1–58.
- [52] Rogers L.J., On function sum theorems connected with the series $\sum_{n=1}^{\infty} \frac{x^n}{n^2}$, *Proc. London Math. Soc. (2)* **4** (1907), 169–189.
- [53] Sergeev S., On pentagon equation, tetrahedron equation, evolution and integrals of motion, arXiv:2303.17730.
- [54] Skandalis G., Operator algebras and duality, in Proceedings of the International Congress of Mathematicians, Vol. I, II (Kyoto, 1990), Mathematical Society of Japan, Tokyo, 1991, 997–1009.
- [55] Stasheff J., How I 'met' Dov Tamari, in Associahedra, Tamari Lattices and Related Structures, Progr. Math., Vol. 299, Birkhäuser, Basel, 2012, 45–63.
- [56] Stasheff J.D., Homotopy associativity of H-spaces. I, Trans. Amer. Math. Soc. 108 (1963), 275–292.
- [57] Street R., The algebra of oriented simplexes, J. Pure Appl. Algebra 49 (1987), 283–335.
- [58] Street R., Fusion operators and cocycloids in monoidal categories, Appl. Categ. Structures 6 (1998), 177–191.
- [59] Suzuki S., The universal quantum invariant and colored ideal triangulations, Algebr. Geom. Topol. 18 (2018), 3363–3402, arXiv:1612.08262.
- [60] Takesaki M., Duality and von Neumann algebras, Bull. Amer. Math. Soc. 77 (1971), 553–557.
- [61] Takesaki M., Duality and von Neumann algebras, in Lectures on Operator Algebras, Lecture Notes in Math., Vol. 247, Springer, Berlin, 1972, 665–779.
- [62] Timmermann T., An invitation to quantum groups and duality. From Hopf algebras to multiplicative unitaries and beyond, EMS Textb. Math., European Mathematical Society (EMS), Zürich, 2008.
- [63] Turaev V.G., Quantum invariants of knots and 3-manifolds, De Gruyter Stud. Math., Vol. 18, Walter de Gruyter & Co., Berlin, 1994.

- [64] Turaev V.G., Viro O.Ya., State sum invariants of 3-manifolds and quantum 6*j*-symbols, *Topology* **31** (1992), 865–902.
- [65] Volkov A.Yu., Beyond the "pentagon identity", Lett. Math. Phys. **39** (1997), 393–397, arXiv:q-alg/9603003.
- [66] Volkov A.Yu., Pentagon identity revisited, Int. Math. Res. Not. 2012 (2012), 4619–4624, arXiv:1104.2267.
- [67] Williams N.J., The first higher Stasheff-Tamari orders are quotients of the higher Bruhat orders, *Electron. J. Combin.* **30** (2023), 1.29, 38 pages, arXiv:2012.10371.
- [68] Wirth J., Stasheff J., Homotopy transition cocycles, J. Homotopy Relat. Struct. 1 (2006), 273–283, arXiv:math.AT/0609220.
- [69] Woronowicz S.L., From multiplicative unitaries to quantum groups, Internat. J. Math. 7 (1996), 127–149.
- [70] Zakrzewski S., Poisson Lie groups and pentagonal transformations, Lett. Math. Phys. 24 (1992), 13–19.